# **Co-opted boards and bidder performance**

Syed Shams

University of Southern Queensland, Toowoomba, Australia Hoa Luong University of Otago, Dunedin, New Zealand, and Nafisa Zabeen Ovi Monash University, Melbourne, Australia China Accounting and Finance Review

Received 5 February 2024 Revised 12 October 2024 Accepted 18 December 2024

#### Abstract

**Purpose** – Defining co-opted directors as those who join a company's board after an incumbent chief executive officer assumes office, this study aims to investigate the influence of co-opted boards on bidder performance. **Design/methodology/approach** – This study applies ordinary least squares regression analyses to a sample of 8,939 acquisition observations announced by US firms spanning the 1999–2019 period. Event study methodology was employed to capture the market response to acquisition announcements. Propensity score matching technique, a two-stage least squares instrumental variable approach and model selection through the Lasso method were performed for robustness and endogeneity correction purposes.

**Findings** – The results depict a significant negative relationship between a co-opted board and return to acquirers, suggesting that managers under co-opted boards make value-destructing Mergers and Acquisitions deals. We also show that the relationship between board co-option and acquisition performance is positively moderated by institutional ownership while being negatively moderated by an entrenched board. Our additional tests reveal that board co-option reduces acquisition efficiency and leads to worse financial performance.

**Practical implications** – This study offers important implications for regulators and policymakers by highlighting how poor monitoring of the board of directors can influence announcement returns.

**Originality/value** – To the best of the authors' knowledge, this paper appears to be the first investigation that makes a link between board co-option and various dimensions of acquisition decision.

Keywords Mergers and acquisitions, Bidder performance, Cumulative abnormal returns, Board co-option **Paper type** Research paper

#### 1. Introduction

Board co-option refers to a situation where directors join the board after the incumbent chief executive officer (CEO) assumes office (Coles, Daniel, & Naveen, 2014). These co-opted directors may exhibit loyalty to the CEO, primarily due to the timing of their nomination, which is influenced by the CEO's authority in the directors' selection process. This loyalty can make them less likely to challenge the CEO's strategies or scrutinise his decisions with the rigor required. Prior studies show that co-opted directors may be weaker monitors because of their close ties to the CEO and their decisions could be swayed by their allegiance to the CEO (Lim, Do, & Vu, 2020; Cook, Chowdhury, & Zhang, 2023). Consequently, a co-opted board can increase CEO power, reduce board independence and create potential conflicts of interest, which can impair the board's ability to make objective and informed decisions. Mergers and Acquisitions (M&A) are significant corporate decisions that are complex and the outcomes carry significant consequences with long-term and far-reaching impacts on stakeholders. Such

#### JEL Classification — G30, G34

© Syed Shams, Hoa Luong and Nafisa Zabeen Ovi. Published in *China Accounting and Finance Review*. Published by Emerald Publishing Limited. This article is published under the Creative Commons Attribution (CC BY 4.0) licence. Anyone may reproduce, distribute, translate and create derivative works of this article (for both commercial and non-commercial purposes), subject to full attribution to the original publication and authors. The full terms of this licence may be seen at http://creativecommons.org/ licences/by/4.0/legalcode



China Accounting and Finance Review Emerald Publishing Limited e-ISSN: 2307-3055 p-ISSN: 1029-807X DOI 10.1108/CAFF-02-2024-0013 decisions require a high degree of independent oversight to assess risks, conduct due diligence and ensure that the transaction benefits shareholders. However, a co-opted board, swayed by allegiance to the CEO, can disrupt this balance, leading to decisions that prioritise the CEO's short-term interests – perhaps empire-building or securing personal financial gains – over the long-term value creation for shareholders. Therefore, a co-opted board exacerbates the agency conflict between management and shareholders in the context of mergers and acquisitions by limiting the board's ability to thoroughly scrutinise potential risk factors and inform stakeholders, including investors and regulators, about governance risks in firms undergoing M&As. Effective board oversight is crucial to ensure that M&A decisions align with shareholder interests rather than being predominantly influenced by the CEO's preferences. Therefore, bidders with high levels of board co-option and decreased board oversight may lead to value-destroying M&A decisions that favour the CEO's personal interests or short-term gains over long-term shareholder value.

From the monitoring perspective of boards, prior literature has documented the negative consequences of board co-option. Due to poor monitoring, co-opted boards decrease turnoverperformance sensitivity (Coles *et al.*, 2014) and dividend payouts (Jiraporn & Lee, 2018) while increase insiders' trading profitability (Rahman, Malik, Ali, & Iqbal, 2021) and risk-taking (Lee, Jiraporn, Kim, & Park, 2021). Co-opted boards are also prone to stock price crashes (Kao, Huang, Fung, & Liu, 2020), corporate misconduct (Zaman, Atawnah, Baghdadi, & Liu, 2021) and restricted debt covenants (Lim *et al.*, 2020). However, the economic consequences of co-option on bidder announcement returns remained unexplored. By investigating how co-opted boards influence M&A outcomes, we uncover the risks posed by weakened board independence and help ensure that future M&A decisions are made with the best interests of shareholders in mind. This study aims to fill this research gap by examining how co-opted boards affect M&A outcomes, contributing to the broader understanding of how board dynamic influences strategic corporate decisions such as mergers and acquisitions decisions.

Our study on board co-option is motivated by the fact that CEOs employ a substantial effect on appointing board members (Coles et al., 2014) whereby boards do not serve as independent monitors due to their allegiance to the CEO. Legally, directors may be independent, but in reality, they may not be independent in monitoring the functions of CEOs (Zaman et al., 2021). Therefore, boards are black boxes, and their composition is compromised when the inner dynamics of boards are unobservable. In essence, the board attributes, for instance, board independence, affect the firm performance, and the social connection between director and CEO can influence the director's influence over CEO monitoring (Hwang & Kim, 2009). The CEOs try to appoint new directors to increase their control over the board [1] and decrease board independence (Hermalin & Weisbach, 1998). The board's monitoring role identifies the board's obligation to compensate managers in such a way that it aligns managers' interests with shareholders (Zattoni, Dedoulis, Leventis, & Ees, 2020). Conventional wisdom maintains that board monitoring is more effective in the presence of more independent directors. However, in practice, CEOs influence board selection in different ways, notably in hiring sympathetic directors (Finkelstein & Hambrick, 1989) or directors with social ties or similar views (Hwang & Kim, 2009). We build on the argument that board co-option creates a poor monitoring environment to examine its impact on M&As, specifically on the bidders' returns. We adopt a stakeholder-agency perspective of the firm and elaborate on value creation through the lens of the board's monitoring function.

According to stakeholder–agency theory, there is an implicit contract between stakeholders (multiple principals) and agents (managers), whereas traditional agency theory focuses on the relationship between shareholders and managers (Hill & Jones, 1992). In the M&A context, directors (whether executive or independent) appointed following a CEO's accession are disinclined to fulfill their fiduciary duty to protect stakeholders by questioning the management's status quo. Humphery-Jenner *et al.* (2022) argue that independent directors, who are arguably less beholden to CEOs, are more effective in disciplining even the powerful

CEOs. Therefore, firms with highly co-opted boards face a governance risk, where the CEO's China Accounting increased control over the board diminishes the board's ability to act independently (Zattoni et al., 2020). This study builds on the argument that board co-option creates a weak governance environment and explores how this dynamic impacts M&A outcomes. By adopting an agency perspective, we investigate how compromised board oversight influences value creation through mergers, with significant implications for investors, managers and regulators.

Following Coles *et al.* (2014), we measure co-option as the percentage of board members hired after the CEO and cumulative abnormal returns (CARs) as the disparity between actual and expected daily returns using Fama, Fisher, Jensen, and Roll (1969) standard event study methodology. Using a sample of US public acquirers from 1999 to 2019, our findings indicate that co-opted boards significantly reduce bidder returns, highlighting the detrimental impact of weakened board monitoring when co-option is present. The results show that public bidders experience significant losses, partly due to a poor monitoring environment. Our results are persistent after controlling for deal and firm-level characteristics.

A potential endogeneity issue is the main concern in establishing a causal relationship between co-option and bidder CARs. We employ a battery of techniques to alleviate endogeneity concerns that are prevalent in corporate research. First, we execute propensity score matching (PSM) regressions on treatment and control samples to address the selection bias arising from firm-related characteristics and bias-omitted variables. Second, we undertake two-stage least squares (2SLS) regressions utilising the board co-option in the earliest year as an instrumental variable (IV) for board co-option in the current year, thus countering the potential reverse causality issue. Third, we employ the machine learning method of the least absolute shrinkage and lasso selection and inference operator to strengthen the baseline regression models. Our baseline results are consistent after controlling for endogeneity-related concerns.

We further reveal the impacts of CEO characteristics including CEO age, CEO risk-taking, CEO tenure and CEO age. We also report that the negative impact of board co-option on acquisition performance is more pronounced when the target board's co-option is higher. Our baseline results are also more pronounced in (1) acquisition of public targets, (2) acquisitions paid by cash, (3) acquisition of targets in related industries, (4) large deals and (5) domestic transactions. Analysing the proposition that the presence of co-opted directors leads to value reduction due to a poor monitoring environment, we conduct subsample analyses on weak- vs. strong-governed firms, proxied by (1) entrenchment index and (2) institutional ownership. Our findings reveal that the relationship between board co-option and acquisition performance is more severe in poorly governed firms while being diminished under the scrutiny of institutional shareholders' co-opted directors.

To further uncover the impact of co-option in M&As, we explore how co-opted boards affect acquisition efficiency: bid premiums, time to complete the deal, partial acquisitions and deal completion. Scholars argue that acquirers' poor takeover performance can be attributed to higher bid premiums (Hussain & Shams, 2022), poor deal assessment (Faleye, Hoitash, & Hoitash, 2011), ignoring partial acquisitions (Nain & Wang, 2018) and failure to complete deals. We find that a higher percentage of co-opted directors on the board reduces the level of acquisition efficiency and subsequently leads to worse performance. The results show that coopted directors (1) pay higher premiums; (2) spend less amount of time, and therefore effort, in assessing the deal; (3) do not prefer partial acquisitions; and (4) are less likely to complete the deal. Our findings consistently demonstrate that co-opted directors fail to make decisions in the interest of shareholders.

This study makes three important contributions to M&A literature. First, we examine the impact of board co-option on bidder announcement returns and suggest that board co-option, among other factors, is detrimental to bidder shareholders. This evidence corroborates the poor monitoring under co-opted boards (Coles et al., 2014; Lim et al., 2020) and highlights that managers under co-opted boards make poor M&A deals, destroying shareholder wealth. We extend the debate over M&As that exhibit acquirers, on average, earning negative returns from

and Finance Review

## CAFR

takeovers (Andrade, Mitchell, & Stafford, 2001; Renneboog & Vansteenkiste, 2019; Starks & Wei, 2013). Second, we add to the literature the role of CEO characteristics, deal characteristics and internal and external monitoring (Morck, Shleifer, & Vishny, 1990; Shleifer & Vishny, 2003; Travlos, 1987) in M&As and identify how these factors can affect the relationship between board co-option and bidder returns. Third, we extend the literature on acquisition efficiency (Humphery-Jenner & Powell, 2011; Hussain & Shams, 2022) and show that bidders with co-opted boards pay higher premiums, take fewer days to complete the deal, do not prefer partial acquisitions and are less likely to complete the deal.

The remainder of this paper is organised as follows: Section 2 discusses the background and develops the hypothesis; Section 3 presents the research design; Section 4 shows analyses and results; and Section 5 concludes the study.

### 2. Literature review and hypothesis development

### 2.1 Board co-option

Existing studies highlight the monitoring role emerging from various firm-level governance mechanisms, including independent directors (Cotter, Shivdasani, & Zenner, 1997), gender diversity (Kirsch, 2018) and board committees (Beasley, Carcello, Hermanson, & Lapides, 2000). However, the contemporary rise in corporate malpractice cases has prompted further research, with recent work identifying directors' associations with executives as a critical reason for weaker monitoring (Baghdadi, Nguyen, & Podolski, 2020; Lim *et al.*, 2020). For instance, Coles *et al.* (2014) demonstrate that a higher percentage of co-opted directors, whether independent or not, reduces the quality of board oversight. This concern aligns with Luong, Minnick, Rivolta, and Sham (2024), who argue that social ties between CEOs and directors undermine board effectiveness.

Recent research also explores the role of CEOs in the selection of directors, raising questions about the rationality and accountability of these appointments (Baghdadi *et al.*, 2020; Jiraporn & Lee, 2018). CEOs may hire directors with whom they share a favourable relationship or similar working ethics (Coles *et al.*, 2014; Wintoki & Xi, 2019). Consequently, co-opted directors often demonstrate loyalty to the CEO, reducing their objectivity in monitoring (Hwang & Kim, 2009). Personal and loyalty ties to management weaken the monitoring capabilities of boards of directors, leading to biased decision-making that often neglects stakeholder interests (Lim *et al.*, 2020).

The economic consequences of co-opted boards have been widely discussed in recent literature. Lartey, Danso, and Boateng (2021) examine the impact of co-option on capital structure, finding that it positively affects financial leverage. Furthermore, Zaman *et al.* (2021) report that co-opted boards contribute to increased incidents of corporate misconduct due to poorly performing board committees. Huang, Han, and Cho (2021) investigate the effects of co-option on risk-taking behaviour, arguing that co-option enhances CEO power in corporate decision-making, which can lead to overestimation of returns on risky projects and, consequently, poor investment decisions. Lim *et al.* (2020) suggest that as the number of co-opted directors rises, creditors demand stricter covenant restrictions. These studies consistently attribute the unfavourable consequences of co-opted boards to their weak monitoring role (Faleye *et al.*, 2011; Chiu, Teoh, & Tian, 2013).

## 2.2 Bidder performance and the role of corporate governance

Prior studies on M&As accentuate the role of corporate governance (see for instance, Cotter *et al.*, 1997; Datta, Iskandar-Datta, & Raman, 2001; Masulis, Wang, & Xie, 2007), proposing that acquirers with higher governance quality earn higher returns and incline to pay lower takeover premiums. These studies also suggest that managers of better-governed firms are good monitors, better estimate takeover synergies and make value-enhancing deals. Black, Jang, and Kim (2006) argue that managers of firms with higher governance standards are less

entrenched and make better investment decisions. Similarly, Doidge, Karolyi, and Stulz China Accounting (2007) proclaim that good governed firms can mitigate agency problems and better perform in M&As. Masulis et al. (2007) find that bidders with more antitakeover provisions (a proxy for poor governance) earn lower returns due to empire building behaviour of bidder managers. while Levi, Li, and Zhang (2014) show that female directors generate higher returns due to less empire-building behaviour. Apart from the positive consequences of good corporate governance, several other studies in M&As show that, on average, bidders perform poorly due to CEO hubris (Roll, 1986), free rider problem (Grossman & Hart, 1980), rational overbidding (Akdogu, 2011), competition in M&A market (Bradley, Desai, & Kim, 1988) and free cash flow (Jensen, 1986).

Another stream of literature has interrogated the impact of board characteristics on the performance of bidding companies. Among other board characteristics, board independence and CEO/Chairman duality are widely discussed in M&A literature to show their impact on the performance of bidder firms. Board independence is considered to be an important factor for effective board monitoring. Independent boards have minimum economic concerns with companies as compared to affiliated boards. Thus, independent boards are perceived as more unbiased and fruitful monitors. For instance, Byrd and Hickman (1992) use tender offers during the 1980s and find that bidder announcement returns are significantly higher in bidding firms where independent outside directors possess at least half of the seats. Masulis et al. (2007) argue that those bidding firms enjoy higher abnormal announcement returns which separate the chairman of the board and CEO positions. Bange and Mazzeo (2004) identify that target shareholders receive higher gains in companies with individuals holding positions of CEO and board chairman. Although the aforementioned studies provide interesting insights into the effect of board characteristics on bidder performance, the impact of board co-option remains unexplored.

#### 2.3 Hypothesis development

The literature has shown that the existence of co-opted boards, irrespective of whether they are independent or executive, erodes their ability to monitor CEOs (Zaman *et al.*, 2021; Coles et al., 2014). Such poor monitoring is linked to the number of co-opted directors, and firms with a greater percentage of co-opted directors not only safeguard CEOs from dismissal (Coles et al., 2014) but also remunerate them with higher stock compensation (Morse, Nanda, & Seru, 2011). Thus, board co-option helps CEOs pursue their personal goals, aggravating agency conflicts between managers and stakeholders.

Stakeholder–agency theory (Hill & Jones, 1992) posits that poorly monitored managers, incentivised by personal motives may engage in activities that not only harm shareholders but also other stakeholders. Stakeholder-agency theory helps explain why co-opted boards enable CEOs to engage in value-destroying decisions. Poor monitoring environments allow CEOs to pursue personal objectives, even when such actions conflict with the welfare of stakeholders. Poorly monitored managers pursue their own motives and engage in malpractice (Jain & Zaman, 2020). For instance, firms with a higher percentage of co-opted directors reserve more cash rather than distributing it to shareholders, suggesting that the presence of excessive cash reserves under a weaker monitoring or governance environment creates an opportunity for managers to pursue short-term agendas. Moreover, "the Airbus Inc. global bribery case" and "Enron accounting fraud" are evidence of a relationship between CEOs and co-opted boards (Zaman et al., 2021).

Building on these discussions, we hypothesise that the propensity of an acquiring firm to engage in a vale-destructing deal will be particularly high in the existence of co-opted board. Co-opted boards, because of their allegiance to CEOs, might be reluctant to fire CEOs in the case of corporate infractions (Coles et al., 2014). Being loval to the CEO, co-opted directors may not maintain proper monitoring and the former may decrease shareholder returns.

and Finance Review Because CEOs in poor monitoring environment can destroy shareholders' wealth, we develop our hypothesis as follows:

*H1*. A higher percentage of co-option can negatively affect bidder cumulative abnormal returns, ceteris paribus.

#### 3. Research design

#### 3.1 Data and sample

We obtain our data from multiple sources. We first extract acquisition data from the Refinitiv Securities Data Corporation's (SDC) Mergers and Acquisition database. We identify 8,939 mergers and acquisitions announced by 1,424 publicly listed unique bidding firms. Target firms include public, private and subsidiaries of domestic and foreign target firms [2]. Following Masulis *et al.* (2007), we keep the mergers and acquisitions announcement that meet the following criteria: (1) both partial and completed acquisitions are included, (2) if the bidding firms acquire more than 50% of the target shares after the acquisition announcements, (3) the deal value disclosed in SDC is more than \$1 million and finally and (4) the acquirer has stock return data (210 trading days prior to acquisition announcements) from the Center for Research in Security Prices Daily Stock Price and Returns file.

Financial data are collected from Compustat, corporate governance data from the BoardEx, institutional investors' ownership data from the FactSet LionShares database, and analysts' forecast data from the Institutional Brokers' Enterprise Systems (I/B/E/S) database. Following Coles *et al.* (2014), we then identify if a director is co-opted if he/she is appointed after the incumbent CEO assumes office to calculate board co-option measure based on the number of co-opted directors on the board. We merge the above data sets and require that all variables have non-missing values to estimate the regression models outlined in Section 3.4 to be included in the sample. We winsorise variables at the top and bottom 1% of the sample distribution or employ a logarithm transformation to avoid the undesirable influence of extreme outliers. Our final sample for the empirical analysis comprises 8,393 observations for the period 1999 to 2019.

Table 1 provides the year-by-year and industry-by-industry [3] distributions of our sample. The number of annual M&A transactions is highest in the years prior to and at the beginning of the global financial crisis (i.e. from 2004 to 2007), implying that many companies were vulnerable to acquisition at the start of this severe economic downturn. A drop in annual M&A transactions can be observed during the global financial crisis and recovery periods to the early-2000.

Regarding industry distribution, to conserve space, we report the top ten bidding industries in Table 1. The Business Services industry constitutes the highest percentage of our total sample at 13.65%, followed by Banking (12.65%), Trading (7.18%) and Electronic Equipment (6.10%) [4]. On the other hand, bidders from the Textiles industry appear to have the lowest number of M&A announcements.

#### 3.2 Variable measurement

3.2.1 Acquirer announcement period returns measures. We employ the conventional event study method based on the market model (Brown & Warner, 1985) and calculate the cumulative abnormal return earned by an acquirer surrounding the announcement day (t = 0) over a three-day event window (from t = -1 to t = +1). In the three-day announcement window, t = 0 is the announcement day identified by the SDC Platinum database. Following prior studies (e.g. Chang, 1998; Masulis *et al.*, 2007; Moeller, Schlingemann, & Stulz, 2004), the firm-specific  $\alpha_i$  and  $\beta_i$  parameters of the market model are estimated using daily returns for the acquirer *i* and for the market index for a 200-day estimation period spanning from t = -210 to t = -11. We exclude the 10-day window around the announcement date from the estimation

| Table 1 | . Sample o  | listribution   |   |              |                 | China Accounting<br>and Finance |
|---------|-------------|--|---|--------------|-----------------|---------------------------------|
| Year    | Obs         | Percent (%)  | Industry category   | Obs          | Percent (%)     | Review                          |
| 1999    | 150         | 1.68   | Business Services   | 1,220        | 13.65           |                                 |
| 2000    | 532         | 5.95   | Banking   | 1,131        | 12.65           |                                 |
| 2001    | 331         | 3.70   | Trading   | 642          | 7.18            |                                 |
| 2002    | 308         | 3.45   | Electronic Equipment                                      | 545          | 6.10            |                                 |
| 2003    | 465         | 5.20   | Pharmaceutical Products                                   | 534          | 5.97            |                                 |
| 2004    | 565         | 6.32   | Petroleum and Natural Gas                                 | 437          | 4.89            |                                 |
| 2005    | 571         | 6.39   | Computers   | 415          | 4.64            |                                 |
| 2006    | 597         | 6.68   | Non-Metallic and Industrial Metal Mining                  | 313          | 3.50            |                                 |
| 2007    | 605         | 6.77   | Medical Equipment   | 301          | 3.37            |                                 |
| 2008    | 478         | 5.35   | Communication   | 285          | 3.19            |                                 |
| 2009    | 381         | 4.26   | Others  | 3,116        | 34.86           |                                 |
| 2010    | 396         | 4.43   |   |              |                 |                                 |
| 2011    | 390         | 4.36   |   |              |                 |                                 |
| 2012    | 400         | 4.47   |   |              |                 |                                 |
| 2013    | 389         | 4.35   |   |              |                 |                                 |
| 2014    | 457         | 5.11   |   |              |                 |                                 |
| 2015    | 488         | 5.46   |   |              |                 |                                 |
| 2016    | 352         | 3.94   |   |              |                 |                                 |
| 2017    | 373         | 4.17   |   |              |                 |                                 |
| 2018    | 388         | 4.34   |   |              |                 |                                 |
| 2019    | 323         | 3.61   |   |              |                 |                                 |
| Total   | 8,939       | 100  | Total   | 8,939        | 100             |                                 |
| bidders | over the pe | e presents sample<br>eriod from 1999 to<br>s' own work | distribution across years (Panel A) and industr<br>o 2019 | ies (Panel I | 3) of US-listed |                                 |

period to minimise the market noise effect because of the common information leaked to the market before the actual announcement date.

3.2.2 Board co-option measures. We follow Coles *et al.* (2014) to define co-opted directors as those who join a company's board after an incumbent CEO assumes office. We use three measures of board co-option. The first measure, *PCO\_OPT*, is calculated by taking the number of co-opted directors divided by the total number of directors on the board. This measure represents the density of co-opted directors on the board and captures the disutility of the board in an effort to monitor CEOs' discretion stemming from the critical role of the CEO in appointing the co-opted directors. The second measure, *NCO\_OPT*, is calculated as the gap between the number of co-opted directors and the total number of directors on the board, representing the severity of board co-option. The third measure, *RCO\_OPT*, is the residual from a regression of *PCO\_OPT* on CEO tenure, which is included to control for the possibility that co-option increases mechanically with CEO tenure and that our models may instead capture the influence of CEO tenure (Coles *et al.*, 2014).

While PCO\_OPT and RCO\_OPT are derived directly from Coles *et al.* (2014), we introduce a new measure of boar co-option, namely NCO\_OPT. This variable adds a new dimension to boar co-option measures by considering the difference between the number of co-opted directors relative to the board size. This approach can reveal a threshold effect, where the presence of co-opted directors might subtly or significantly influence board decisions once a certain threshold is crossed. Specifically, we extend the implication of the critical mass theory beyond the traditional gender diversity application to suggest that there is a critical point at which the influence of co-opted directors becomes more pronounced (Dahlerup, 2006; Oliver & Marwell, 2001). By capturing this relative measure, NCO\_OPT allows for a more nuanced assessment of how co-option might impact board dynamics and decision-making

CAFR

processes, providing insights into the conditions under which situations co-opted directors significantly shape board outcomes.

3.2.3 Control variables. We employ three sets of control variables that are known to be determinants of acquirer announcement returns and board co-option in the literature. First, both acquirer announcement returns and board co-option are influenced by a range of firm characteristics. Previous studies find that the market reacts to the acquirers' profitability, firm size and cash holding (Luong, Gunasekarage, & Shams, 2021; Moeller *et al.*, 2004; Rosen, 2006). Accordingly, we control for firm size (*SIZE*), cash holding (*CASH*) and profitability (*ROA*). Likewise, investors consider acquirers' debt-to-asset ratio and market valuation to make acquisition decisions (Humphery-Jenner & Powell, 2011; Masulis *et al.*, 2007). Consequently, we use firm leverage (*LEV*) and Tobin's Q (*TOBINQ*) as control variables. Recent anecdotal evidence suggests that the number of analysts following influences acquirer abnormal returns (Wright, Kroll, & Elenkov, 2002). This variable has also been identified by the literature as being related to board co-option given its powerful external monitoring mechanism (Brennan & Subrahmanyam, 1995), allowing us to control for *LOGNANALYSTS* in our model.

Our second set of control variables is included to address the concern of the possibility that our results might be due to CEO and board characteristics rather than board co-option. These variables are the tenure of the CEO (*CEO\_TENURE*) and the CEO and chair dual position (*CEO\_DUALITY*), the size of the board of directors (*BOARD\_SIZE*) and the percentage of independent directors (*BOARD\_INDEP*). We control for the impact of CEO tenure, which is one source of managers' power (Finkelstein, 1992) because CEOs with long tenure could strengthen CEOs' influence on the management structure and board composition (Hermalin & Weisbach, 1998). Moreover, it is likely that powerful CEOs may have preferences in the recruitment of directors with whom they are connected (Fracassi & Tate, 2012), leading to board co-option (Coles *et al.*, 2014). CEO tenure has also been found to be associated with acquisition performance (Zhou, Dutta, & Zhu, 2020). Furthermore, the combination of the CEO and chair position can reflect the power vested in the CEO, which can heighten the possible conflicts of interest and reduce the level of boards' monitoring function (Davidson, Goodwin-Stewart, & Kent, 2005). Similarly, the literature also reports that board size and board independence can impact acquisition performance (Masulis *et al.*, 2007).

Third, it is well established in prior scholarship that acquirer performance is influenced by deal characteristics. In particular, it is likely that stock price predicts the potential synergy based on the relative size of the acquirer and target (Alexandridis, Fuller, Terhaar, & Travlos, 2013; Moeller *et al.*, 2004) and the industry in which they operate (Morck *et al.*, 1990; Shleifer & Vishny, 2003). Therefore, we control for relative size (*RELATIVESIZE*) and acquirer and target industry (*RELATEDID*). Since the method of payment (Travlos, 1987), the presence of competing bidders (De, Fedenia, & Triantis, 1996) and the type of offer (i.e. hostile or tender) (Schwert, 2000; Pablo, 2013) signal the market about the bargaining power between acquirer and target, we include cash payment (CASH), stock payment (STOCK), competing bidder (*COMPETINGBID*) and tender offer (*TENDEROFFER*) in our regression. Detailed definitions of the variables are presented in the Appendix.

#### 3.3 Analytical model

To examine the effect of board co-option on acquirer performance, we estimate the following regression model:

$$CAR_{i,t} = \propto_0 + \beta_1(COOP_{i,t-1}) + \sum_{i} \beta_i Controls_{i,t-1} + \sum_{i} \beta_i Year_{i,t} + \sum_{i} \beta_i Industry_{i,t} + \varepsilon_{i,t}$$
(1)

where *i* and *t* refer to firm and year, respectively. The dependent variable is acquirer's abnormal announcement returns (*CAR*). The independent variable is the proportion of co-opted directors

(PCO OPT) or number of co-opted directors (NCO OPT). Control variables consist of firm China Accounting size (SIZE), leverage ratio (LEV), cash holding (CASH), Tobin's Q (TOBINO), return-onassets ratio (ROA), the natural logarithm of the number of analysts following (LOGNANALYSTS) and relative size which is the transaction value divided by the market value of the acquirer one month prior to the acquisition announcement (RELATIVESIZE). CEO characteristics and board specifics include the CEO-chair title concentration (CEO DUALITY), natural logarithm of the year-length of CEO term (CEO TENTURE), natural logarithm of the number of directors on the board (LNBOARD SIZE) and the percentage of independent directors on the board (BOAR INDEP). Other controls are dummy variables that take a value of one if deals are paid solely by cash (ALLCASH) or by stock (STOCK), if the bidder and target are from the same industries (RELATEDID) if the deal is tender (TENDEROFFER) and if there are at least two acquirers making an offer to a target (COMPETINGBID). In all regressions, we include industry and year fixed effects and report *t*-statistics with robust standard errors clustered at the firm level. Year fixed effects allow us to control for common macroeconomic factors, whereas industry fixed effects account for all time-invariant industry-level characteristics which might be correlated with both the level of acquirer performance and board co-option.

#### 3.4 Descriptive statistics and correlations

Table 2 presents the descriptive statistics for the variables used in our main analyses. Panel A reports our main dependent variable. The mean value of acquirer performance, CAR, is 0.0020. This positive mean value indicates that, on average, the sample acquirers earn positive abnormal returns of 0.2% during the three days surrounding the announcement date. Board cooption measures, PCO OPT and NCO OPT on average, have means of 0.4615 and 4.6197 and 0, respectively, meaning that nearly a half of the directors are appointed after the incumbent CEOs assume office and that there is a difference of five between the number of co-opted directors and the total number of directors the on the board.

Panel B shows the summary statistics of bidder characteristics. The average natural logarithm of firms' assets in our sample is 7.9, generated from the un-tabulated average total assets of \$2.84 billion. The average leverage ratio and cash holding of the acquiring firm are 0.2328 and 0.1483, respectively, indicating that a typical firm in our sample finances 23.28% of its assets by debt and maintains 14.83% of cash holding. As reflected by a Tobin's Q of 2.0136, the sample comprises growing firms that possess future growth opportunities valued by the market who generate a positive profitability performance of 1.45% return on assets. A typical firm in our sample is followed by 10 analysts (reported as a corresponding natural logarithm of 1.9856) prior to the deal announcement.

Turning to corporate governance in Panel C, more than half of the CEO (53.07%) also hold a Chair position. A typical CEO's tenure is four years and an average board has about 10 members (which translate into logarithm value of 1.5474 and 2.3973, respectively). More than half of the board (68.24%) are independent directors. Panel D reports the descriptive statistics for deal characteristics. On average, the relative size of the deal to market capitalisation of the bidder is 19.41%; 31.25% of the deals are financed solely in cash and 14.07% are paid by stock. We find that acquiring a target in a related industry is quite common (52.95%), whereas only 6.48% of the deals are tender offers and 2.36% of the acquisitions attract competing bidders.

Table 3 reports the results of the correlation matrix. We find that acquirer performance is negatively and significantly correlated with both measures of board co-option. Such findings underpin and support our hypothesis that the market does not react favourably to takeovers conducted by firms with a high proportion of co-opted directors on the board. The correlations among the three board co-option variables (PCO OPT, NCO OPT and RCO OPT) are above 0.89 and board size and firm size variables, BOARD SIZE and SIZE, have a high correlation of 0.62, which is consistent with expectations in the literature. The correlations among the

and Finance Review

|                        | Mean              | Std. Dev.   | 1st quartile | Median    | 3rd quartile |
|------------------------|-------------------|-------------|--------------|-----------|--------------|
| Panel A: dependent ar  | d independent var | righter     | *            |           | 1            |
| CAR                    | 0.0020            | 0.0621      | -0.0252      | 0.0006    | 0.0265       |
| PCO OPT                | 0.4615            | 0.3543      | 0.1250       | 0.4545    | 0.7333       |
| NCO OPT                | 4.6176            | 3.9859      | 1.0000       | 4.0000    | 7.0000       |
| RCO_OPT                | -0.0000           | 0.3440      | -0.2883      | 0.0084    | 0.2542       |
| Panel B: firm characte | pristics          |             |              |           |              |
| SIZE                   | 7.9114            | 2.1402      | 6.4928       | 7.9058    | 9.3755       |
| TA (\$mil)             | 28400.8691        | 131142.7569 | 660.3410     | 2712.8980 | 11795.3110   |
| LEV                    | 0.2328            | 0.1904      | 0.0739       | 0.2052    | 0.3525       |
| CASH                   | 0.1483            | 0.1666      | 0.0302       | 0.0826    | 0.2046       |
| TOBINQ                 | 2.0136            | 1.5765      | 1.1086       | 1.5001    | 2.2547       |
| ROA                    | 0.0145            | 0.1707      | 0.0064       | 0.0353    | 0.0835       |
| Panel C: corporate go  | vernance          |             |              |           |              |
| CEO_DUALITY            | 0.5307            | 0.4991      | 0.0000       | 1.0000    | 1.0000       |
| LNCEO_TENURE           | 1.5474            | 0.5997      | 1.0986       | 1.6094    | 1.9459       |
| CEO_TENURE             | 4.6406            | 3.6037      | 2.0000       | 4.0000    | 6.0000       |
| LNBOARD_SIZE           | 2.3973            | 0.2905      | 2.1972       | 2.3979    | 2.5649       |
| BOARD_SIZE             | 10.4724           | 3.4645      | 8.0000       | 10.0000   | 12.0000      |
| BOARD_INDEP            | 0.6824            | 0.1680      | 0.5833       | 0.7143    | 0.8000       |
| LNANALYSTS             | 1.9956            | 1.0008      | 1.3863       | 2.0794    | 2.7726       |
| NUM_ANALYSTS           | 10.0168           | 8.9549      | 3.0000       | 7.0000    | 15.0000      |
| Panel D: deal charact  | eristics          |             |              |           |              |
| RELATIVESIZE           | 0.1941            | 0.3655      | 0.0120       | 0.0576    | 0.1990       |
| ALLCASH                | 0.3125            | 0.4635      | 0.0000       | 0.0000    | 1.0000       |
| ALLSTOCK               | 0.1407            | 0.3478      | 0.0000       | 0.0000    | 0.0000       |
| RELATEDID              | 0.5295            | 0.4992      | 0.0000       | 1.0000    | 1.0000       |
| TENDEROFFER            | 0.0648            | 0.2461      | 0.0000       | 0.0000    | 0.0000       |
| COMPETINGBID           | 0.0236            | 0.1518      | 0.0000       | 0.0000    | 0.0000       |

**Note(s):** This table presents the descriptive statistics for the main dependent and independent variables in Panel A, acquirer-level financial control variables in Panel B, corporate governance in Panel C and deal characteristics in Panel D. We provide variables definitions in the Appendix **Source(s):** Authors' own work

remaining variables are in a small magnitude, signifying that multicollinearity is not a main issue of the models (Gujarati & Porter, 2009).

## 4. Analyses and results

4.1 Board co-option and market reactions to acquisition announcements

Table 4 presents the results of regressing the board co-option measures on the acquisition performance proxy. We find that acquirer abnormal announcement returns (*CAR*), in all models, are negatively and statistically related to the board co-option measures. Specifically, the coefficient estimates on acquisition performance are negative and significant at the level of 1% in Columns (1)–(3). The results are not only statistically significant but also economically significant. For example, the coefficient estimate reported for acquirer performance in Column (1), which is the model specification using *PCO\_OPT* as a proxy for board co-option, is -0.0071. This coefficient estimate suggests that a one-standard-deviation increase in *PCO\_OPT* leads to a reduction in *CAR* of 0.0004 ( $-0.0071 \times 0.0621$ ). Considering that the average acquirer announcement return is only 0.0020, the baseline results reported in Table 4 are,

Table 3. Correlation matrix

|  |             | (1)                        | (2)   | (3)   | (4)  | (5)   | (6)                        | (7)                          | (8)                         | (9)                    | (10)                     | (11)   | (12)                         | (13)                   | (14)                          | (15)                          | (16)             | (17)                 | (18)    | (19)     | (20)   |
|--|-------------|----------------------------|---|---|--|---|----------------------------|------------------------------|-----------------------------|------------------------|--------------------------|--|------------------------------|------------------------|-------------------------------|-------------------------------|------------------|----------------------|---------|----------|--------|
| CAR  | (1)         | 1.00                       |   |   |  |   |                            |                              |                             |                        |                          |  |                              |                        |                               |                               |                  |                      |         |          |        |
| PCO_OPT  | (2)         | $-0.05^{***}$              | 1.00  |   |  |   |                            |                              |                             |                        |                          |  |                              |                        |                               |                               |                  |                      |         |          |        |
| NCO_OPT  | (3)         | $-0.06^{***}$              | $0.91^{***}$                                | 1.00  |  |   |                            |                              |                             |                        |                          |  |                              |                        |                               |                               |                  |                      |         |          |        |
| RCO_OPT  | (4)         | $-0.04^{***}$              | $0.97^{***}$                                | $0.89^{***}$                                | 1.00   |   |                            |                              |                             |                        |                          |  |                              |                        |                               |                               |                  |                      |         |          |        |
| SIZE   | (5)         | $-0.08^{+++}$              | $0.16^{***}$                                | 0.36***                                     | $0.20^{***}$                                 | 1.00  |                            |                              |                             |                        |                          |  |                              |                        |                               |                               |                  |                      |         |          |        |
| LEV  | (6)         | 0.05***                    | -0.01                                       | 0.01  | -0.00  | 0.21***                                       | 1.00                       |                              |                             |                        |                          |  |                              |                        |                               |                               |                  |                      |         |          |        |
| CASH   | (7)         | $-0.03^{***}$              | -0.00                                       | $-0.09^{***}$                               |  | -0.27***                                      | -0.33****                  | 1.00                         |                             |                        |                          |  |                              |                        |                               |                               |                  |                      |         |          |        |
| TOBINQ   | (8)         | 0.01                       | 0.02*                                       | $-0.02^{**}$                                | -0.00  | $-0.07^{***}$                                 |                            | 0.38***                      | 1.00                        |                        |                          |  |                              |                        |                               |                               |                  |                      |         |          |        |
| ROA  | (9)         | 0.02                       | $0.09^{***}$                                | 0.13***                                     | 0.12***                                      | 0.30  | 0.01                       | $-0.18^{***}$                | 0.09***                     | 1.00                   |                          |  |                              |                        |                               |                               |                  |                      |         |          |        |
| CEO_DUALITY  | (10)        |                            | -0.01                                       | 0.01  | 0.01   | 0.12****                                      | 0.07***                    | $-0.05^{***}$                | 0.00                        | 0.05****               | 1.00                     |  |                              |                        |                               |                               |                  |                      |         |          |        |
| LNCEO_   | (11)        | 0.01                       | $-0.26^{***}$                               | $-0.23^{***}$                               | $-0.04^{***}$                                | $0.10^{***}$                                  | 0.06***                    | $-0.09^{***}$                | $-0.12^{***}$               | 0.11***                | 0.09***                  | 1.00   |                              |                        |                               |                               |                  |                      |         |          |        |
| TENURE   |             | 0.00****                   | 0.1=***                                     | 0. 1=***                                    | o  | 0.00***                                       | 0.0.4***                   | 0.0=***                      | 0.40***                     | o * * *                |                          | o o .***                                     |                              |                        |                               |                               |                  |                      |         |          |        |
| LNBOARD_SIZE   |             |                            | 0.17****                                    | 0.45****                                    | 0.17***                                      | 0.62***                                       | 0.04***                    | -0.25***                     | -0.12****                   | 0.14****               | 0.02                     | -0.04****                                    | 1.00                         |                        |                               |                               |                  |                      |         |          |        |
| BOARD_INDEP  | (13)        | 0.00                       | 0.09****                                    | 0.13****                                    | 0.17 <sup>****</sup><br>0.18 <sup>****</sup> | 0.15  | $-0.02^{*}$<br>$0.02^{**}$ | $-0.02^{*}$ $0.05^{***}$     | -0.10****<br>0.27***        | 0.11***                | 0.11***                  | 0.37 <sup>****</sup><br>0.04 <sup>****</sup> | 0.09                         | 1.00                   |                               |                               |                  |                      |         |          |        |
| LNANALYSTS   |             | -0.07***                   | 0.17 <sup>***</sup><br>-0.05 <sup>***</sup> | 0.24 <sup>***</sup><br>-0.09 <sup>***</sup> | 0.18   | 0.57 <sup>****</sup><br>-0.18 <sup>****</sup> | 0.02<br>0.13***            | -0.05                        | $-0.13^{***}$               | 0.26***                | $0.16^{***}$<br>-0.03*** | -0.04  | 0.30                         | 0.10<br>-0.05***       | 1.00<br>-0.20***              | 1 00                          |                  |                      |         |          |        |
| RELATIVESIZE   |             | 0.02**                     | -0.05<br>$0.06^{***}$                       | -0.09<br>$0.07^{***}$                       | -0.07***<br>0.08***                          | -0.18<br>$0.08^{***}$                         | 0.13<br>0.03**             |                              |                             | -0.15***<br>0.12***    | -0.03                    | -0.05<br>$0.06^{***}$                        | -0.11                        | -0.05<br>$0.10^{***}$  | -0.20<br>$0.10^{***}$         |                               | 1.00             |                      |         |          |        |
| ALLCASH<br>ALLSTOCK  | (16)        | $0.03^{**}$<br>-0.08^{***} | $-0.05^{***}$                               | -0.07                                       | 0.08<br>-0.08***                             | -0.08   | $-0.11^{***}$              | 0.02<br>0.04 <sup>****</sup> | $-0.02^{*}$<br>$0.06^{***}$ | -0.12<br>$-0.17^{***}$ | $0.01 \\ -0.02$          | $-0.14^{***}$                                | 0.00<br>0.07 <sup>****</sup> | -0.10<br>$-0.16^{***}$ | $-0.05^{***}$                 | $-0.14^{***}$<br>$0.12^{***}$ | 1.00<br>-0.27*** | 1.00                 |         |          |        |
| RELATEDBID   | (17) (18)   |                            | -0.05<br>-0.01                              | -0.04<br>0.01                               | -0.08<br>-0.01                               | -0.04<br>$-0.03^{**}$                         | -0.11<br>$-0.03^{**}$      | -0.13 <sup>***</sup>         | -0.11***                    | -0.17<br>-0.01         | 0.05***                  | 0.03***                                      | 0.00***                      | -0.16<br>$0.02^{**}$   | -0.03<br>$-0.07^{***}$        | 0.12                          | -0.27            | 0.00***              | 1.00    |          |        |
| TENDEROFFER  | (10) $(19)$ |                            | -0.01<br>$0.06^{***}$                       | 0.01  | 0.05***                                      | 0.09***                                       | -0.05                      | -0.15                        | 0.00                        | -0.01<br>$0.04^{****}$ | -0.05<br>$0.02^{*}$      | $-0.03^{***}$                                | 0.05                         | 0.02                   | -0.07<br>0.09 <sup>****</sup> | $-0.02^{*}$                   | 0.20****         | $-0.05^{***}$        | -0.01   | 1.00     |        |
| COMPETINGBID   | (20)        | $-0.03^{*}$                | 0.04***                                     | 0.03***                                     | 0.03   | 0.03***                                       | 0.03***                    | 0.01                         | -0.00                       | 0.04                   | 0.02**                   | 0.01   | 0.03                         | 0.00                   | 0.04***                       | -0.02<br>$0.10^{***}$         | 0.20             | 0.01                 | 0.03*** | 0.13***  | * 1.00 |
| <b>Note(s):</b> This ta<br>significance at<br><b>Source(s):</b> Au | 10%         | , 5% or                    | 1%, res                                     |   | 0  | ariables                                      | ised in tl                 | ne subse                     | quent an                    | alyses. A              | All varia                | bles are o                                   | defined i                    | n the <mark>Ap</mark>  | pendix.                       | The aste                      | erisk *, *       | <sup>**</sup> Or *** | denote  | s statis | stical |

|  | (1)<br>CAP                       | (2)<br>CAD                       | (3)                                   |
|--|----------------------------------|----------------------------------|---------------------------------------|
|  | CAR                              | CAR                              | CAR                                   |
| PCO_OPT                                      | $-0.0071^{***}$<br>(-3.44)       | -                                | -                                     |
| NCO_OPT                                      | _                                | $-0.0007^{***}$<br>(-3.57)       | _                                     |
| <br>RCO_OPT                                  | -                                | _                                | $-0.0064^{***}$<br>(-3.12)            |
| SIZE   | $-0.0015^{***}$<br>(-2.74)       | $-0.0014^{***}$<br>(-2.63)       | $-0.0015^{***}$<br>(-2.77)            |
| LEV  | 0.0129** (2.55)                  | 0.0129** (2.54)                  | 0.0129** (2.55)                       |
| CASH   | $(-0.0143^{**})$<br>(-2.48)      | $(-0.0144^{**})$<br>(-2.50)      | $-0.0143^{**}$<br>(-2.48)             |
| TOBINQ                                       | (-2.46)<br>0.0028***<br>(4.71)   | (-2.30)<br>0.0028***<br>(4.70)   | (-2.46)<br>0.0028***<br>(4.71)        |
| ROA  | 0.0073 (1.21)                    | 0.0071<br>(1.19)                 | 0.0072 (1.20)                         |
| CEO_DUALITY                                  | 0.0012<br>(0.78)                 | 0.0013<br>(0.83)                 | 0.0012 (0.80)                         |
| LNCEO_TENURE                                 | $(-0.0036^{**})$<br>(-2.41)      | $(-0.0036^{**})$<br>(-2.35)      | $-0.0026^{*}$<br>(-1.86)              |
| LNBOARD_SIZE                                 | 0.0007<br>(0.21)                 | 0.0031 (0.88)                    | 0.0007<br>(0.21)                      |
| BOARD_INDEP                                  | (0.21)<br>-0.0015<br>(-0.28)     | (0.00)<br>-0.0014<br>(-0.26)     | (0.21)<br>-0.0018<br>(-0.33)          |
| LNANALYSTS                                   | (-0.20)<br>-0.0027***<br>(-2.71) | (-0.20)<br>-0.0027***<br>(-2.74) | (-0.33)<br>$-0.0027^{***}$<br>(-2.73) |
| RELATIVESIZE                                 | 0.0020<br>(0.63)                 | 0.0019<br>(0.63)                 | 0.0020 (0.63)                         |
| ALLCASH                                      | 0.0023 (1.61)                    | 0.0023<br>(1.62)                 | 0.0023                                |
| ALLSTOCK                                     | $(-0.0116^{***})$<br>(-4.74)     | $(-0.0117^{***})$<br>(-4.78)     | $-0.0116^{***}$<br>(-4.73)            |
| RELATEDBID                                   | 0.0004<br>(0.26)                 | 0.0004 (0.25)                    | 0.0004 (0.26)                         |
| TENDEROFFER                                  | -0.0057**<br>(-2.25)             | -0.0057**<br>(-2.24)             | -0.0057**<br>(-2.25)                  |
| COMPETINGBID                                 | -0.0048<br>(-1.07)               | (-1.10)                          | -0.0048<br>(-1.08)                    |
| CONSTANT                                     | 0.0128<br>(0.93)                 | 0.0070 (0.50)                    | 0.0087<br>(0.62)                      |
| Year fixed effects<br>Industry fixed effects | Yes<br>Yes                       | Yes<br>Yes                       | Yes<br>Yes                            |
| $R^2$<br>N                                   | 0.0385<br>8,939                  | 0.0384<br>8,939                  | 0.0382<br>8,939                       |

**Note(s):** This table presents the regression results of co-opted boards on acquisition performance with other control variables. Robust two-tailed *t*-statistics clustered by firm are presented in parentheses. The superscripts \*\*\*\*, \*\* and \* correspond to statistical significance at the 1%, 5% and 10% levels, respectively. We present the variable definitions in the Appendix **Source(s):** Authors' own work

therefore, both statistically and economically significant. The results are consistent with the prediction that board co-option reduces abnormal returns surrounding the announcement date.

For other variables – deals initiated by large acquirers and acquirers with a high level of China Accounting cash holding, greater analyst coverage, paying the deals solely by stock, offering tender deals and being conducted by long-tenured CEOs seem to be responded to negatively by the market. By contrast, highly leveraged and highly valuated acquirers are likely to outperform over the announcement period. These results are generally consistent with prior literature (e.g. Luong et al., 2021; Masulis et al., 2007; Moeller et al., 2004; Rosen, 2006; Schwert, 2000).

and Finance Review

#### 4.2 Endogeneity correction

Our results suggest thus far that board co-option negatively relates to acquirer performance. We now consider endogeneity issues that are prevalent because of the black-box nature of the corporate boardroom (Wintoki, Linck, & Netter, 2012). First, although using stock market reactions reduces the potential reverse causality, it may still be present when high-performing acquisitions could change the board co-option. For example, the existing evidence suggests that acquisition performance affects CEOs' ability to build networks through firm reputation (Hoang & Antoncic, 2003), altering CEOs' ability to attract and influence the recruitment of directors in the CEOs' connection. Second, despite the fact that we have included a range of control variables in our model, there is a possibility that we may have omitted some variables that might mechanically impact this relationship (Larcker & Rusticus, 2010). For example, some unidentified factors, such as executives' personal and demographic backgrounds (Jensen & Zajac, 2004) or organisational culture (Davidson, Dey, & Smith, 2015; Schoenberg, 2000), may affect both recruitment strategies and acquisition decisions. Finally, a possibility exists that the relation between our variables in our model can be problematic because CEO attributes and board characteristics, such as CEO duality, CEO tenure, board size and board independence, can drive both board co-option and acquisition performance (Bertrand & Schoar, 2003; Finkelstein, Hambrick, & Cannella, 2009; Gunasekarage, Luong, & Truong, 2020; Zhou *et al.*, 2020), allowing one to believe that the acquisition returns are potentially the outcome of CEO and board characteristics rather than board co-option. To address the above endogeneity concerns, we follow prior studies (Wintoki et al., 2012; Wooldridge, 2010) and adopt three sophisticated techniques -PSM, IV 2SLS regressions and Lasso selection and inference models - to explore if our contention holds un endogeneity assumptions.

4.2.1 Propensity score matching approach. As a further cross-check of the baseline results. we employ a PSM analysis to address both the selection bias arising from firm-related characteristics and bias-omitted variables (Rosenbaum & Rubin, 1983). To execute our PSM analysis, we define firms whose proportion of board co-option is above the industry and year median as the treatment group. The control firms are those whose ratio of board co-option is below the industry and year median value.

To ensure that our treated and control firms are comparable, we match treatment and control firms using PSM, where we utilise the nearest neighbour with replacement matching using a calliper of 0.01. In the first stage, the matching procedure is undertaken based on all the control variables that are used in the baseline regressions in Table 4. We report the results in Table 5. Panel A of Table 5 reports the univariate mean comparisons between treatment and control firms' characteristics and their corresponding t-statistics. The results demonstrate that the average values of the matching variables are qualitatively the same across the treatment and control firms. We then perform PSM regressions using the post-matched sample in the second stage reported in Panel B of Table 5. The coefficients of board co-option measures remain negative and significant at the 1% level. These findings are consistent with our baseline results that show the presence of co-opted directors lowers the announcement returns of acquirers. The results indicate that our baseline finding that board co-option aggravates acquisition performance is not confounded by potential self-selection bias and omitted variables of the regression residuals.

4.2.2 Instrumental variables. In this subsection, we adopt the IV approach to address the potential endogeneity concerns. We follow Li, Gong, Zhang, and Koh (2018) to use board

| $O_{I} \Pi \Pi$ |
|-----------------|
|-----------------|

**Table 5.** Co-opted board and market reactions to acquisition announcements: propensity score matching analysis

| MeanMeanValueMeanMeanValueMeanMeanMeanMeanMeanValueSIZE7,82377,87230.357,94748,00300.307,75507,78160.55LEV0.23270.23090.690.23460.23430.960.23430.960.23430.960.23430.960.14920.15190.51COBINQ1.91021.98930.231.97161.95340.641.96772.05330.02CAA0.01240.01620.340.01920.01950.9330.01140.01840.162ECO_DUALITY0.52710.55260.52100.730.52890.52510.77NECAC1.57081.53580.151.54501.53430.491.58301.56110.13TENURENBOARD_SIZE2.39562.40490.192.42122.42830.312.38362.39200.24SIGARJ_INDEP0.686110.68460.720.68840.69010.710.67860.67890.93NANALYSTS1.96532.00240.122.02802.04430.531.92550.1290.12ALLASTOCK0.14100.13420.420.55860.54020.740.53240.5110.42ALLSTOCK0.14100.13420.53580.54020.740.53240.5100.44CO_OPT  |                              | High<br>DPCO_<br>OPT<br>(treated<br>group)  | Low<br>DPCO<br>OPT<br>(contro<br>group) | bl  | High<br>DNCO_<br>OPT<br>(treated<br>group)       | Low<br>DNCC<br>OPT<br>(contro<br>group) | ol   | High<br>DRCO_<br>OPT<br>(treated<br>group) | Low<br>DRCO_<br>OPT<br>(control<br>group) | _                           |
|---|------------------------------|---|---|---|--|---|--|--|---|-----------------------------|
| EV         0.2327         0.2309         0.69         0.2346         0.2343         0.96         0.2371         0.2259         0.16           CASH         0.1503         0.1489         0.74         0.1434         0.1432         0.66         1.9677         2.0533         0.22           COA         0.0124         0.0162         0.34         0.0192         0.0135         0.33         0.5249         0.5210         0.35         0.5289         0.5211         0.75           CNCEO_         1.5758         1.5358         0.15         1.5450         1.5343         0.49         1.5830         1.63           NBOARD_SIZE         2.3956         2.4049         0.19         2.4212         2.4283         0.31         2.3836         2.3920         0.24           QAARD_INDEP         0.68611         0.6846         0.72         0.6884         0.6001         0.71         0.6786         0.6789         0.93           NANALYSTS         1.9553         2.0024         0.12         2.0248         0.914         0.18         0.1219         0.124         0.2125         0.1299         0.12           ALLCASH         0.3009         0.3277         0.20         0.2338         0.5402         0.74  |                              | Mean  | Mean                                    | <i>p-</i><br>value                                | Mean   | Mean                                    | <i>p</i> -<br>value                          | Mean                                       | Mean                                      | <i>p</i> -<br>value         |
| CASH         0.1503         0.1489         0.74         0.1434         0.1432         0.96         0.1492         0.1519         0.51           IOBINQ         1.9102         1.9803         0.23         1.9716         1.9534         0.64         1.9677         0.0533         0.023           ROA         0.0124         0.0152         0.34         0.0122         0.95         0.33         0.5251         0.75           NCEO         1.5708         1.5358         0.15         1.5450         1.5343         0.49         1.5800         1.5611         0.73           CACD         1.5708         1.5358         0.15         1.5450         1.5343         0.49         1.5800         1.5611         0.73           CARD         NDARD_SIZE         2.3956         2.4049         0.19         2.4212         2.4283         0.31         2.3836         0.5921         0.74         0.736         0.93           SOARD_INVESIZE         0.2108         0.1959         0.11         0.2048         0.1914         0.18         0.2155         0.1384         0.14           ALLCASH         0.300         0.327         0.239         0.340         0.10         0.291         0.525         0.521 <td< td=""><td>SIZE</td><td>7.8237</td><td>7.8723</td><td>0.35</td><td>7.9474</td><td>8.0030</td><td>0.30</td><td>7.7550</td><td>7.7816</td><td>0.59</td></td<> | SIZE                         | 7.8237  | 7.8723                                  | 0.35  | 7.9474   | 8.0030                                  | 0.30   | 7.7550                                     | 7.7816                                    | 0.59                        |
| FOBINQ       1.9102       1.9893       0.23       1.9716       1.9534       0.64       1.9677       2.0533       0.26         ROA       0.0124       0.0162       0.34       0.0195       0.33       0.0114       0.0184       0.16         ECD_DUALITY       0.5271       0.5526       0.940       0.5245       0.5120       0.35       0.5289       0.5251       0.75         NEOARD_SIZE       2.3956       2.4049       0.19       2.4212       2.4283       0.31       2.3836       2.3920       0.42         SNBOARD_INDEP       0.68611       0.6846       0.72       0.6884       0.6901       0.71       0.6763       0.93         NANALYSTS       1.9653       2.0024       0.12       2.0280       2.0443       0.53       1.9263       1.9399       0.15         SILATIVESIZE       0.2108       0.1579       0.11       0.0248       0.1914       0.18       0.2155       0.1328       0.125       0.3228       0.121       0.142         ALLSTOCK       0.309       0.3277       0.20       0.239       0.3400       0.15       0.532       0.529       0.57         COMPETINGBID       0.5253       0.5293       0.74       0.5324   | LEV                          | 0.2327  | 0.2309                                  | 0.69  | 0.2346   | 0.2343                                  | 0.96   | 0.2371                                     | 0.2259                                    | 0.16                        |
| ROA         0.0124         0.0162         0.34         0.0192         0.0195         0.93         0.0114         0.0184         0.184           ECO_DUALITY         0.5271         0.5262         0.94         0.5245         0.5120         0.35         0.5289         0.5251         0.75           NCEO_         1.5708         1.5358         0.15         1.5450         1.5343         0.49         1.5830         1.5611         0.13           CENURE         2.3956         2.4049         0.19         2.4212         2.4283         0.31         2.3866         2.3920         0.24           OARD_INDEP         0.68611         0.684         0.6901         0.71         0.6780         0.33           SALATVESIZE         0.2108         0.1579         0.10         0.2939         0.3400         0.10         0.2955         0.3228         0.15           ALLCASH         0.3009         0.327         0.20         0.339         0.0667         0.664         0.24         0.555         0.1384         0.14           RELATIVESIZE         0.324         0.000         0.0157         0.0336         0.00         0.0211         0.0291         0.04           TENDEROFTER         0.620         0.6  | CASH                         | 0.1503  | 0.1489                                  | 0.74  | 0.1434   | 0.1432                                  | 0.96   | 0.1492                                     | 0.1519                                    | 0.51                        |
| ECO_DUALITY         0.5271         0.5262         0.94         0.5245         0.5120         0.35         0.5289         0.5251         0.75           NCCO_         1.5708         1.5358         0.15         1.5450         1.5343         0.49         1.5830         1.5611         0.13           IENURE  | fobinq                       | 1.9102  | 1.9893                                  | 0.23  | 1.9716   | 1.9534                                  | 0.64   | 1.9677                                     | 2.0533                                    | 0.26                        |
| NCEO_       1.5708       1.5358       0.15       1.5450       1.5343       0.49       1.5830       1.5611       0.13         VRBOARD_SIZE       2.3956       2.4049       0.19       2.4212       2.4283       0.31       2.3836       2.3920       0.24         30ARD_INDEP       0.68611       0.6846       0.72       0.6884       0.6901       0.71       0.6786       0.6789       0.93         NANALYSTS       1.9653       2.0024       0.12       2.0280       2.0443       0.53       1.5256       0.3228       0.12         ALLASH       0.3009       0.3277       0.20       0.2939       0.3400       0.10       0.2955       0.3228       0.15         ALLASTOCK       0.1410       0.1342       0.42       0.5566       0.1299       0.15       0.1555       0.3348       0.14         KELATTEDBID       0.5253       0.5239       0.740       0.5386       0.5402       0.74       0.5326       0.529       0.57         COMPETINGBID       0.127       0.0324       0.00       0.0157       0.0336       0.00       0.0211       0.291       0.0467         Parel A: first stage       Panel A: first       Panel A: first       Panel A: first   | ROA                          | 0.0124  | 0.0162                                  | 0.34  | 0.0192   | 0.0195                                  | 0.93   | 0.0114                                     | 0.0184                                    | 0.18                        |
| NBOARD_SIZE       2.3956       2.4049       0.19       2.4212       2.4283       0.31       2.3836       2.3920       0.24         SOARD_INDEP       0.68611       0.6846       0.72       0.6884       0.6901       0.71       0.6786       0.6786       0.939       0.15         NANALYSTS       1.9653       2.0024       0.12       2.0280       2.0443       0.153       1.9263       1.9939       0.11         RELATIVESIZE       0.2108       0.1959       0.11       0.2048       0.1914       0.18       0.2155       0.1329       0.12         ALLSTOCK       0.1110       0.3277       0.20       0.2393       0.3400       0.10       0.2955       0.3228       0.12         XLSTOCK       0.1110       0.324       0.42       0.1299       0.15       0.1555       0.1384       0.44         RELATEDBID       0.5253       0.5293       0.74       0.5358       0.5402       0.74       0.5326       0.067       0.0684       0.24       0.0596       0.629       0.57         COMPETINGBID       0.0172       0.0324       0.00       0.0157       0.0336       0.00       0.0211       0.0211       0.024         Panel B: propensit//secmathar fir  | LNCEO_                       |   |   |   |  |   |  |  |   | 0.75<br>0.13                |
| BOARD_INDEP         0.68611         0.6846         0.72         0.6884         0.6901         0.71         0.6786         0.6789         0.93           NANALYSTS         1.9653         2.0024         0.12         2.0280         2.0443         0.53         1.9939         0.15           SELATIVESIZE         0.2108         0.1959         0.11         0.2048         0.10         0.2955         0.3228         0.15           ALLCASH         0.3009         0.3277         0.20         0.2939         0.3400         0.10         0.2955         0.3228         0.15           ALLSTOCK         0.1410         0.1342         0.42         0.1566         0.1299         0.15         0.1555         0.1384         0.14           RELATIEDBID         0.5253         0.5293         0.74         0.5358         0.5402         0.74         0.5324         0.0629         0.57           COMPETINGBID         0.0172         0.0324         0.00         0.0157         0.0336         0.00         0.0211         0.0291         0.04           Panel A: first stage         Panel A: first stage         Panel A: car         -(-2.85)         -(-2.83)           CO_OPT         -         -         -         -(-2.85)   |                              | 2 2056  | 2 1010                                  | 0.10  | 2 4212   | 2 1283                                  | 0.31   | 2 2826                                     | 0 2020                                    | 0.24                        |
| NANALYSTS       1.9653       2.0024       0.12       2.0280       2.0443       0.53       1.9263       1.9939       0.15         RELATIVESIZE       0.2108       0.1959       0.11       0.2048       0.1914       0.18       0.2155       0.1929       0.12         ALLCASH       0.3009       0.3277       0.20       0.2939       0.3400       0.10       0.2955       0.3228       0.15         ALLSTOCK       0.1410       0.1342       0.42       0.156       0.1299       0.15       0.1555       0.1384       0.14         RELATEDBID       0.5253       0.5293       0.74       0.5358       0.5402       0.74       0.5324       0.5410       0.44         COMPETINGBID       0.0172       0.0324       0.00       0.0157       0.0336       0.00       0.0211       0.0291       0.04         Panel B: propensity score matching regression results       Panel R: first stage       Panel B: second stage       (1)       (2)       (3)       (4)       (5)       (6)         DPCO_       DNCO_       DRCO_       (-2.86)       -       -       -       (-2.86)       -       -       (-2.45)         SIZE       0.0149***       0.0084       0.0182***   | _                            |   |   |   |  |   |  |  |   |                             |
| RELATIVESIZE         0.2108         0.1959         0.11         0.2048         0.1914         0.18         0.2155         0.1929         0.12           LLCASH         0.3009         0.3277         0.20         0.2939         0.3400         0.10         0.2955         0.3228         0.12           LLSTOCK         0.1410         0.1342         0.42         0.1566         0.1299         0.14         0.5555         0.3228         0.540           KLATEDBID         0.5253         0.5293         0.74         0.5358         0.5402         0.74         0.5356         0.6629         0.57           COMPETINGBID         0.0172         0.0324         0.00         0.0157         0.0336         0.00         0.211         0.291         0.04           COMPETINGBID         0.0172         0.0324         0.00         0.0157         0.0336         0.00         0.211         0.291         0.04           Panel A: first stage         Panel B: second stage         (1)         (2)         (3)         (4)         (5)         (6)           PCO_OPT         -         -         -         -         -         (-2.45)         (-2.45)           SIZE         0.0149***         0.0084         0.  |                              |   |   |   |  |   |  |  |   |                             |
| ALLCASH       0.3009       0.3277       0.20       0.2939       0.3400       0.10       0.2955       0.3228       0.15         ALLSTOCK       0.1410       0.1342       0.42       0.1566       0.1299       0.15       0.1555       0.1384       0.14         RELATEDBID       0.5253       0.5293       0.674       0.5358       0.5402       0.74       0.5324       0.5060       0.629       0.55         COMPETINGBID       0.0172       0.0324       0.00       0.0157       0.0336       0.00       0.0211       0.0291       0.044         Panel B: propensity       score matching regression results       Panel B: second stage       (1)       (2)       (3)       (4)       (5)       (6)         PCO_OPT       -       -       -       -       (-2.86)       -       -       (-2.45)         CO_OPT       -       -       -       -       -       -       (-2.45)       -       -       (-2.45)         SIZE       0.0149***       0.0084       0.0182***       -0.0016**       -0.0011       -0.0016**         CO_OPT       -       -       -       -       -       -       (-2.45)       -       -       -       <  |                              |   |   |   |  |   |  |  |   |                             |
| ALLSTOCK       0.1410       0.1342       0.42       0.1566       0.1299       0.15       0.1555       0.1384       0.14         RELATEDBID       0.5253       0.5293       0.74       0.5358       0.5402       0.74       0.5324       0.5410       0.46         COMPETINGBID       0.0172       0.0324       0.00       0.0157       0.0336       0.00       0.0211       0.0291       0.04         Panel B: propensity score matching regression results         Panel B: propensity score matching regression results       Panel B: second stage       (1)       (2)       (3)       (4)       (5)       (6)         DPCO_       DNCO_       DRCO_       OPT       CAR       CAR       CAR         VCO_OPT       -       -       -       -0.0068***       -       -       -         VCO_OPT       -<   |                              |   |   |   |  |   |  |  |   |                             |
| RELATEDBID         0.5253         0.5293         0.74         0.5358         0.5402         0.74         0.5324         0.5410         0.48           TENDEROFFER         0.0620         0.0672         0.39         0.0607         0.0684         0.24         0.0596         0.0629         0.57           COMPETINGBID         0.0172         0.0324         0.00         0.0157         0.0336         0.00         0.0211         0.0291         0.042           Panel A: first stage         Panel A: first stage         Panel B: second stage         (1)         (2)         (3)         (4)         (5)         (6)           DPCO_         DNCO_         DRCO_         OPT         OPT         CAR         CAR         CAR           VCO_OPT         -         -         -         -         -         (-2.83)         - <td></td>   |                              |   |   |   |  |   |  |  |   |                             |
| $\begin{array}{cccccccccccccccccccccccccccccccccccc$  |                              |   |   |   |  |   |  |  |   |                             |
| COMPETINGBID         0.0172         0.0324         0.00         0.0157         0.0336         0.00         0.0211         0.0211         0.0211         0.0211         0.044           Panel B: propensity score matching regression results         Panel A: first stage         Panel B: second stage         (1)         (2)         (3)         (4)         (5)         (6)           DPCO_         DNCO_         DRCO_         ORCO_         ORCO_         (-2.86)         -         -           VCO_OPT         - </td <td></td>  |                              |   |   |   |  |   |  |  |   |                             |
| Panel B: propensity score matching regression results           Panel A: first stage         Panel B: second stage           (1)         (2)         (3)         (4)         (5)         (6)           DPCO_         DNCO_         DRCO_         DRCO_         OPT         CAR         CAR         CAR           PCO_OPT         -         -         -         -         (-2.86)         -         -           NCO_OPT         -         -         -         -         (-2.83)         -         -           RCO_OPT         -         -         -         -         (-2.45)         -         -           SIZE         0.0149***         0.0084         0.0182***         -0.0016**         -0.0011         -0.0010           LEV         0.0154         0.0358         0.0152         0.0140***         0.0011**         0.0150***         0.011**           CASH         0.0337         0.0505         0.0469         -0.0147***         -0.0082         -0.016***           SIZE         0.0137         0.5055         0.0469         -0.0147***         -0.0082         -0.016***           CASH         0.0337         0.0505         0.0469         -0.0147***   |                              |   |   |   |  |   |  |  |   |                             |
| DPCO_<br>OPT         DNCO_<br>OPT         DRCO_<br>OPT         DRCO_<br>OPT         CAR         CAR         CAR           PCO_OPT         -         -         -0.0068***         -         -         -           NCO_OPT         -         -         -         -0.0007***         -         -           NCO_OPT         -         -         -         -         -         -         -           NCO_OPT         -         -         -         -         -         -         -         -           RCO_OPT         - <td< th=""><th>Panel B: propensity</th><th>Panel A</th><th>: first st</th><th>age</th><th></th><th></th><th></th><th>0</th><th>(6)</th><th></th></td<>  | Panel B: propensity          | Panel A   | : first st                              | age   |  |   |  | 0  | (6)                                       |                             |
| $\begin{array}{cccccccccccccccccccccccccccccccccccc$  |                              | DPCO_   | Ì                                       | NCO_  | DRCO_  |   | ,  |  | . ,                                       |                             |
| $\begin{array}{llllllllllllllllllllllllllllllllllll$  | PCO_OPT                      | _   | _                                       |   | _  |   |  | _  | _   |                             |
| $\begin{array}{cccccccccccccccccccccccccccccccccccc$  |                              |   |   |   |  | (-                                      | -2.86)                                       | 0.000                                      | <b>-</b> +++                              |                             |
| $\begin{array}{cccccccccccccccccccccccccccccccccccc$  | NCO_OPT                      | -   | -                                       |   | -  | -                                       |  |  |   |                             |
| $\begin{array}{cccccccccccccccccccccccccccccccccccc$  |                              |   |   |   |  |   |  | (-2.83)                                    |   | 0057**                      |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$   | CO_OP1                       | -   | _                                       |   | _  | _                                       |  | _  |   |                             |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$   |                              | 0.01.40*  | ** 0                                    | 0004  | 0.0100*  | 4 4                                     | 0.0016**                                     | 0.001                                      |   |                             |
| $\begin{array}{cccccccccccccccccccccccccccccccccccc$  | DIZE                         |   |   |   |  |   |  |  |   |                             |
| $\begin{array}{cccccccccccccccccccccccccccccccccccc$  | LEV                          | 0.0154  | Ó                                       | .0358   | 0.0152   | Ó                                       | .0140***                                     | 0.0150 <sup>*</sup>                        | *** <b>0.</b> 0                           | 101**                       |
| $\begin{array}{cccccccccccccccccccccccccccccccccccc$  | 7ASH                         | 0.0337  | Ó                                       | .0505   | 0.0469   | _                                       | -0.0147**                                    | -0.008                                     | 2 –0                                      | .0167**                     |
| $\begin{array}{cccccccccccccccccccccccccccccccccccc$  | JAJII                        |   |   |   | -0.0031  | Û Û                                     | .0023***                                     | 0.0024*                                    | *** 0.0                                   | 025***                      |
| CEO_DUALITY         0.0142         0.0065         0.0136         0.0014         0.0018         0.0012           (1.35)         (0.54)         (1.30)         (0.82)         (0.98)         (0.70)           LNCEO_TENURE         -0.1874***         -0.1919***         -0.2127***         -0.0023         -0.0025         -0.0030*           LNBOARD_SIZE         -0.0042         -0.1562***         -0.0038         0.0021         0.0028         -0.0000  | TOBINQ                       | -0.0034   |   | -0.59)  | (-0.74)  |   |  |  |   |                             |
| $\begin{array}{cccccccccccccccccccccccccccccccccccc$  | TOBINQ                       | -0.0034<br>(-0.81)<br>0.0515*   | * 0                                     | .0726***  | 0.0396*  | Ò                                       | .0096*                                       |  |   |                             |
| NBOARD_SIZE         -0.0042         -0.1562***         -0.0038         0.0021         0.0028         -0.0000  | FOBINQ<br>ROA                | -0.0034<br>(-0.81)<br>0.0515*<br>(2.46)<br>0.0142   | ()<br>** 0<br>()<br>()<br>()            | .0726***<br>2.73)<br>.0065<br>0.54)               | 0.0396*<br>(1.89)<br>0.0136<br>(1.30)            | 0<br>(1<br>0<br>((                      | .0096*<br>1.77)<br>.0014<br>).82)            | (2.43)<br>0.0018                           | (1.)<br>0.0<br>(0.)                       | 94)<br>012<br>70)           |
|   | fobinq<br>Roa<br>Ceo_duality | $\begin{array}{r} -0.0034 \\ (-0.81) \\ 0.0515* \\ (2.46) \\ 0.0142 \\ (1.35) \\ -0.1874 \end{array}$ | (<br>* 0<br>(i<br>0<br>(!<br>4*** -     | .0726***<br>2.73)<br>.0065<br>0.54)<br>-0.1919*** | 0.0396*<br>(1.89)<br>0.0136<br>(1.30)<br>-0.2127 | 0<br>(1<br>0<br>((<br>7*** _            | .0096*<br>I.77)<br>.0014<br>).82)<br>-0.0023 | (2.43)<br>0.0018<br>(0.98)<br>-0.002       | (1.)<br>0.0<br>(0.)<br>5 -0               | 94)<br>012<br>70)<br>.0030* |

#### Table 5. Continued

China Accounting and Finance Review

| Panel B: propensity    | score matching | g regression res | ults      |                     |            |                     | Reviev |
|------------------------|----------------|------------------|-----------|---------------------|------------|---------------------|--------|
|                        | Panel A: firs  | st stage         |           | Panel B: seco       | ond stage  |                     |        |
|                        | (1)            | (2)              | (3)       | (4)                 | (5)        | (6)                 |        |
|                        | DPCO_          | DNCO_            | DRCO_     |                     |            |                     |        |
|                        | OPT            | OPT              | OPT       | CAR                 | CAR        | CAR                 |        |
| BOARD_INDEP            | 0.0540*        | 0.0588*          | 0.0444    | -0.0048             | -0.0035    | -0.0034             |        |
|                        | (1.71)         | (1.72)           | (1.48)    | (-0.82)             | (-0.55)    | (-0.58)             |        |
| LNANALYSTS             | 0.0227***      | 0.0285***        | 0.0242*** | -0.0029**           | -0.0040*** | -0.0034***          |        |
|                        | (3.79)         | (4.28)           | (4.20)    | (-2.52)             | (-3.17)    | (-3.04)             |        |
| RELATIVESIZE           |                | -                | -         | 0.0042 <sup>*</sup> | 0.0021     | 0.0030 <sup>´</sup> |        |
|                        |                |                  |           | (1.86)              | (0.86)     | (1.36)              |        |
| ALLCASH                | -              | -                | -         | 0.0018              | 0.0005     | 0.0033*             |        |
|                        |                |                  |           | (0.97)              | (0.23)     | (1.79)              |        |
| ALLSTOCK               | _              | _                | _         | -0.0113***          | -0.0114*** | -0.0109***          |        |
|                        |                |                  |           | (-4.50)             | (-4.29)    | (-4.55)             |        |
| RELATEDBID             | -              | -                | -         | 0.0002              | -0.0003    | 0.0005 <sup>´</sup> |        |
|                        |                |                  |           | (0.12)              | (-0.18)    | (0.29)              |        |
| TENDEROFFER            | -              | -                | -         | -0.0027             | -0.0034    | -0.0052             |        |
|                        |                |                  |           | (-0.81)             | (-0.93)    | (-1.56)             |        |
| COMPETINGBID           | -              | -                | -         | -0.0065             | -0.0064    | -0.0090*            |        |
|                        |                |                  |           | (-1.26)             | (-1.15)    | (-1.79)             |        |
| CONSTANT               | 0.0635         | 0.4850***        | 0.3755*** | 0.0696*             | 0.0099     | -0.0007             |        |
|                        | (0.60)         | (3.04)           | (3.74)    | (1.79)              | (0.21)     | (-0.03)             |        |
| Year fixed effects     | Yes            | Yes              | Yes       | Yes                 | Yes        | Yes                 |        |
| Industry fixed         | Yes            | Yes              | Yes       | Yes                 | Yes        | Yes                 |        |
| effects                |                |                  |           |                     |            |                     |        |
| Wald chi square        | 949.9656       | 1683.9419        | 904.9965  | 0.0434              | 0.0441     | 0.0407              |        |
| Pseudo $R^2/\hat{R}^2$ | 0.0773         | 0.1392           | 0.0730    |                     |            |                     |        |
| Ν                      | 8,939          | 8,939            | 8,939     | 6,480               | 5,464      | 6,802               |        |

**Note(s):** This table presents the PSM results of co-opted board and acquisition performance with other control variables. Panel A reports the mean differences between the control group and matched group. Panel B reports the regression estimates using these two groups. The superscripts \*\*\*, \*\* and \* correspond to statistical significance at the 1%, 5% and 10% levels, respectively. We present the variable definitions in the Appendix **Source(s):** Authors' own work

co-option in the earliest year (*PCO\_OPT\_FIRST*, *NCO\_OPT\_FIRST* and *RCO\_OPT\_FIRST*) as the IVs. We refer to Larcker and Rusticus (2010) to identify and verify a valid IV that satisfies three major criteria, namely relevance, exclusion restriction and exogeneity.

First, the relevance condition requires that the IV is correlated with the respective endogenous regressor (i.e. *PCO\_OPT*, *NCO\_OPT* and *RCO\_OPT*). Empirically, a firm's board structure is relatively stable overtime (Graham, Kim, & Leary, 2017) with board independence exhibits persistence and path-dependence (Hermalin & Weisbach, 1998). Therefore, board dynamics can influence directors' tenure and initial board co-option is likely to continue, at least partially, into the subsequent years, which provides a strong correlation with the board co-option in the concurrent year.

Second, to satisfy the exclusion restriction criterion, it is unlikely that our IV, the initial board co-option, will not have a direct effect on the acquisition decisions in subsequent years. Clearly, the board's composition in its earliest formation occurs before the particular acquisition decisions in the concurrent year. Since the initial board co-option establishes a foundational structure, it is expected to influence later decisions indirectly, primarily through its effect on board independence and monitoring capabilities (Coles *et al.*, 2014; Zaman *et al.*, 2021). Therefore, the initial board co-option serves as a valid instrument because it meets the exclusion restriction criterion by ensuring that its effects on the dependent variable, acquisition

performance, occur solely through its established relationship with board dynamics rather than direct influence.

Third, the exogeneity criterion requires that the IV must not be correlated with the error term in the model, ensuring that any unobserved factors affecting the dependent variable are unrelated to the IV. In our setting, since this initial co-option occurs prior to the specific acquisition decisions being made in subsequent years, it is unlikely to be influenced by unobserved factors that might affect the firm's acquisition decisions later. Furthermore, the firm's board structure, especially its initial co-option state, reflects historical governance dynamics rather than immediate business decisions (Larcker & Rusticus, 2010; Jiraporn & Lee, 2018). This temporal separation between the board co-option and acquisition decisions reduces the risk of reverse causality or omitted variable bias, ensuring that the initial board co-option is not correlated with the error term. Thus, the exogeneity assumption holds, as the IV influences acquisition outcomes only through its effect on board dynamics, not through any direct correlation with the unobservable factors captured in the model's error term (Larcker & Rusticus, 2010; Coles *et al.*, 2014).

We report the results in Table 6. In all regressions, we include industry and year fixed effects and report *t*-statistics with robust standard errors clustered at the firm level. In Panel A, we present the first-stage regression results using three measures of board co-option, *PCO\_OPT*, *NCO\_OPT* and *RCO\_OPT* as the dependent variables and the instruments as the main independent variables in Columns (1)–(3). The coefficient estimates on *PCO\_OPT\_FIRST*, *NCO\_OPT\_FIRST* and *RCO\_OPT\_FIRST* in the three first-stage regressions are positive and significant at the 1% level, suggesting that our IVs are positively associated with board co-option. The *p*-value of the Kleibergen-Paap rk Wald *F* statistic of the first-stage regressions is significant at the 1% level, rejecting the null hypothesis of weak identification (Larcker & Rusticus, 2010). Therefore, the coefficient estimates and their corresponding *t*-statistics in the second stage are likely to be unbiased and inferences based on them are reasonably valid. We then present the second-stage regression results in Columns (4)–(6). The coefficient estimates on the instrumented values of board co-option are negative and statistically significant at the 1% level across all columns. The IV regression results are consistent with our baseline findings and further support our predictions that board co-option reduces acquisition performance.

4.2.3 Model selection through the Lasso method. Regression model used in this study has certain limitations due to its high dimensionality, while explanatory variables can be prone to multicollinearity problem. As a result, it would be difficult to test individual regression coefficients of independent and control variables due to inflated standard errors. Accordingly, we implement the least absolute shrinkage and selection operator (Lasso) to strengthen the regression model by checking the validity of selected variables. This method provides a more robust analysis that allows finding important variables in a large set of potential determinants (Tibshirani, 1996; Belloni, Chernozhukov, Fernández-Val, & Hansen, 2017). By using the Lasso method, we can make the results easier to interpret and resolve the problem of multicollinearity since Lasso shrinks regression coefficients by penalising their magnitude and provides a narrow set of important variables.

Panel A of Table 7 presents the results of the Lasso selection model for board co-option and acquisition performance. Three types of Lasso model selection methods (adaptive, cross-validation and plug-in) are used to estimate effects for potential independent and control variables to be included in the model. We observe that both independent and control variables are similar in terms of their coefficients to baseline regression. Most of the explanatory variables also hold their respective coefficient signs in the Lasso selection model. Therefore, the selection of variables is well justified and does not significantly affect the impact of board co-option on acquisition performance.

We should also note that Lasso models are inherently selection models. This group of models select covariates and estimates coefficients without providing standard errors. However, several modified versions of the Lasso model allow for deriving standard errors of estimates. Specifically, the double-selection Lasso method uses selected control variables in

 Table 6.
 Endogeneity-corrected regression output

China Accounting and Finance

| uu | I II. | unce |
|----|-------|------|
|    | De    | VIDW |
|    |       |      |

| PCO_OPT_FIRST<br>PCO_OPT_FIRST0.6017***<br>RCO_OPT_FIRSTNCO_OPT_FIRST<br>RCO_OPT_FIRST-0.0523***<br>(23.00)RCO_OPT_FIRST<br>PCO_OPT_FIRST-0.0523***<br>(23.00)RCO_OPT_FIRST<br>PCO_OPT(40.40)IVPCO_OPT<br>PCO_OPT(-3.39)<br>(-3.07)INCO_OPT<br>PCO_OPTIVRCO_OPT<br>PCO_OPTIVRCO_OPT<br>PCO_OPTSIZE0.0149***<br>(0.55)0.01520.0129***<br>(0.254)-0.0017***<br>(-3.07)CASH0.0337<br>(0.55)0.01520.0129***<br>(0.014)-0.0141**<br>(-0.0141**TOBINQ-0.0351<br>(-0.0131(0.250)(2.57)<br>(-2.44)(2.44)<br>(-2.44)CCA(0.355)<br>(0.257**0.0365<br>(0.00530.0062<br>(0.0028***ROA0.0515**<br>(-0.214)(-2.30)<br>(-2.44)(-2.44)<br>(0.024***ROA0.0515**<br>(-1.651)(-2.30)<br>(-1.615)(-1.45)<br>(-1.651)NBOARD_INDEP0.0540*<br>(0.54)0.0386<br>(0.56)0.0033<br>(0.021<br>(-0.022)0.0021<br>(0.022***ROA0.0560*<br>(-1.651)(-2.41)<br>(-1.651)(-1.65)<br>  |                          | Panel A: first (1) | stage (2)   | (3)         | Panel B: seco (4) | ond stage<br>(5) | (6)        | and Finance<br>Review |
|---|--------------------------|--------------------|-------------|-------------|-------------------|------------------|------------|-----------------------|
| (27.28)         -         -         -         -         -           RCO_OPT_FIRST         -         -         (40.40)         -         -         -           IVPCO_OPT         -         -         (40.40)         -         -         -           IVPCO_OPT         -         -         -         (-3.30)         -         -           IVRCO_OPT         -         -         -         (-3.00)         -         -           IVRCO_OPT         -         -         -         -         (-3.00)         -           IVRCO_OPT         -         <   |                          |                    |             |             |                   |                  |            |                       |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$   | PCO_OPT_FIRST            |                    | -           | -           | -                 | -                | -          |                       |
| RC0_OPT_FIRSTIVPC0_OPT0.0091***IVNC0_OPTIVRC0_OPTIVRC0_OPTIVRC0_OPTIVRC0_OPTIVRC0_OPTIVRC0_OPTIVRC0_OPTIVRC0_OPTIVRC0_OPTIVRC0_OPTIVRC0_OPTIVRC0_OPTIVRC0_OPTIVRC0_OPT  | NCO_OPT_FIRST            | . ,                |             | -           | -                 | -                | -          |                       |
| IVPCO_OPT         -         -         -0.0091***         -         -           IVNCO_OPT         -         -         -0.0087***         -           IVRCO_OPT         -         -         -         -         -           IVRCO_OPT         -         -         -         -         -         -           IVRCO_OPT         -         -         -         -         -         -           SIZE         0.0149****         0.0084         0.0182***         -0.0016***         -0.0017***         -0.0017***           G334)         (.55)         (.414)         (-3.07)         (-3.07)         (-3.07)           LEV         0.0154         0.0358         0.0152         0.014**         -0.0141**         -0.0141**           (0.83)         (1.21)         (1.21)         (-2.40)         (-2.44)         (-2.44)           TOBINQ         -0.0034         -0.0029         -0.0031         0.0028***         0.0028***           (0.81)         (-0.72)         (0.355         0.136         0.0013         0.012         0.012           CASH         (.424)         -0.0024         -0.152***         -0.0034         0.0021         -0.0020           LEV   | RCO_OPT_FIRST            | -                  |             |             | -                 | -                | -          |                       |
| IVNCO_OPT         -         -         -         -0.0087***         -           IVRCO_OPT         -         -         -         -         -0.0094***           IVRCO_OPT         -         -         -         -0.0017***           SIZE         0.0149***         0.0084         0.0182***         -0.0016***         -0.0017***           (3.34)         (1.55)         (4.14)         (-3.04)         (-3.07)         -         -           LEV         0.0137         0.0505         0.0469         -0.014**         -0.0114**         -0.011***           (ASH)         0.037         0.0505         0.0469         -0.014**         -0.014**         -0.014**           (DAST)         0.055         (1.21)         (-2.50)         (-2.44)         (-2.44)           TOBINQ         -0.0034         -0.029         -0.031         0.0028***         0.0028***         0.0028**           ROA         0.0515**         0.0726***         0.0380         0.0065         0.0063         0.00612         0.0012           LNEO_TENURE         -1645         0.0726***         0.0380         0.0011         -0.0020         -0.0020           LNEO_TENURE         -0.1629         (-1617)         (-157   | IVPCO_OPT                | -                  | -           | _           |                   | -                | -          |                       |
| IVRCO_OPTSIZE0.0149***0.00840.0182***-0.0016***-0.0017***-0.0017***(3.34)(1.55)(4.14)(-3.04)(-3.07)(-3.07)(-3.07)LEV0.01540.0380.01520.103**00.0129**0.0129**(0.56)(1.17)(0.56)(2.57)(2.54)(2.54)CASH0.0370.0500.0469-0.0144**-0.0141**(0.89)(1.21)(1.21)(-2.50)(-2.44)(-2.44)TOBINQ-0.0034-0.0029-0.00310.0028***0.0028***(0.81)(-0.55)(-1.45)(0.79)(0.79)ROA(2.46)(2.70)(1.36)(1.30)(1.04)(1.46)(2.46)(2.77)(1.36)(1.09)(1.04)CEO_DUALITY0.01420.0050.00550.00630.0062LINCEO_TENURE-0.187***-0.127***-0.0210.0012-0.0201LINCEO_TENURE-0.187***-0.127***-0.021-0.0020-0.0020LINBOARD_SIZE-0.018**0.03800.0041-0.0021-0.0021LINBOARD_SIZE0.054*0.028***0.0242***-0.0228**-0.0228**(1.17)(1.27)(1.48)(-0.47)(-0.47)(-0.47)LINBOARD_SIZE0.052***-0.0230.00230.00230.0224***(1.60)0.50*0.038*0.044-0.00380.00120.01  | IVNCO_OPT                | -                  | -           | -           | _                 |                  | -          |                       |
| SIZE       0.0149***       0.0084       0.0182***       -0.0016***       -0.0017***       -0.0017***         LEV       0.154       0.338       0.152       0.1030**       0.0129***       0.0129***         LEV       0.056       (1.17)       0.56)       (2.57)       (2.54)       (2.54)         CASH       0.037       0.0505       0.0469       -0.0141**       -0.0141**       -0.0141**         TOBINQ       -0.0034       -0.0029       -0.0031       0.0028***       0.0028***       0.0028***         CASH       0.0515**       0.076***       0.0306*       0.0012       0.014***       0.0012         CEO_DUALITY       0.042       0.065       0.013       0.0012       0.0012         LICEO_TENURE       0.147***       0.013**       0.012       0.0012         LINBOARD_SIZE       -0.0042       -0.152***       -0.0021       -0.0020       -0.0020         LINBOARD_SIZE       -0.0042       -0.152***       -0.0021       -0.0021       -0.0021         LINBOARD_SIZE       -0.0042       -0.562**       -0.0021       -0.0026       -0.0028**         LINBOARD_SIZE       0.027***       0.028***       0.024***       -0.0028***       -0.0026***  | IVRCO_OPT                | -                  | -           | -           | -                 | -                |            |                       |
| LEV0.01540.03580.01520.0139**0.0129**(0.56)(1.7)0.56)(2.57)(2.54)(2.54)CASH0.03370.05050.0469-0.0141**-0.0141**(0.89)(1.21)(1.21)(-2.50)(-2.44)(-2.44)TOBINQ(-0.031)(-0.029)-0.00210.0028***0.0028***(0.081)(-0.59)(-0.74)(4.80)(4.74)(4.74)ROA0.0515**0.0726**0.036*0.00650.00630.0062CD_DUALITY0.01420.00500.0130.0120.0012(1.64)(1.73)(1.89)(1.09)(1.04)(1.04)CEO_DUALITY0.0142-0.0141**-0.0021-0.0020-0.0020LNCEO_TENURE-0.1874***-0.1919**-0.021-0.0021-0.0020-0.0020LNCEO_TENURE-0.1874***-0.1527**-0.0023-0.0020-0.0021LNBOARD_SIZE-0.0042-0.1552**-0.0023-0.0021-0.0011(1.63)(1.72)(1.48)(-0.64)(-0.47)(-0.47)LNAALYSTS0.022***0.028***-0.0028**-0.0028**-0.0028**ALLCASH(0.62)(1.59)(1.59)ALLCASH(0.62)(2.48)(-4.89)RELATEDBID(0.25)(0.34)(0.34)COMPETINGBID(0.025**-0.0056**- <td< td=""><td>SIZE</td><td></td><td></td><td></td><td></td><td></td><td>-0.0017***</td><td></td></td<>   | SIZE                     |                    |             |             |                   |                  | -0.0017*** |                       |
| CASH         0.0337         0.0505         0.0469         -0.0144**         -0.0141**         -0.0141**           TOBINQ         -0.034         -0.0029         -0.031         (-2.44)         (-2.44)           TOBINQ         -0.0034         -0.0029         -0.0028***         0.0028***         0.0028***           ROA         (0.515**         0.0726***         0.0396*         0.0065         0.0065         0.0062           (2.46)         (2.73)         (1.89)         (1.09)         (1.04)         (1.04)           CEO_DUALITY         0.0142         0.0065         0.013         0.0012         0.0012           (1.35)         (0.54)         (1.30)         (0.82)         (0.79)         (0.79)           LNCEO_TENURE         -0.1874***         -0.019***         -0.0021         -0.0020         -0.0020           LNBOARD_SIZE         -0.042         -0.1582***         -0.0034         -0.0025         -0.0021           LNBOARD_INDEP         0.052**         0.0242***         -0.0028***         -0.0028***         -0.0028***           LNANALYSTS         0.0227***         0.0242***         -0.0028***         -0.0028***         -0.0028***           ALLCASH         -         -         -         <  | LEV                      | 0.0154             | 0.0358      | 0.0152      | 0.0130**          | 0.0129**         | 0.0129**   |                       |
| TOBINQ         -0.0034         -0.0031         0.0028***         0.0028***         0.0028***           ROA         0.0515***         0.0726***         0.0063         0.0062           CA         0.0515***         0.0726***         0.0012         0.0063           CEO_DUALITY         0.0142         0.005         0.013         0.012         0.012           LNCEO_TENURE         -0.1874***         -0.1919***         -0.2127***         -0.0021         -0.0020         -0.0020           LNBCARD_SIZE         -0.0424         -0.1517         (-1.55)         (-1.51)         (-1.55)         (-1.45)           LNBOARD_INDEP         0.0042         -0.1562***         -0.0031         0.0025         -0.0021           LNAALYSTS         0.0227***         0.0285***         0.024**         -0.0024         -0.0025           LNAALYSTS         0.0227***         0.2285***         0.0242***         -0.0024         -0.0025           LLSTOK         -         -         -         0.0023         0.0023         0.0023           ALLCSTOK         -         -         -         0.0024***         -0.0121***         -0.0120***           ALLSTOCK         -         -         -         0.0225*** <td< td=""><td>CASH</td><td>0.0337</td><td>0.0505</td><td>0.0469</td><td>-0.0144**</td><td><math>-0.0141^{**}</math></td><td>-0.0141**</td><td></td></td<> | CASH                     | 0.0337             | 0.0505      | 0.0469      | -0.0144**         | $-0.0141^{**}$   | -0.0141**  |                       |
| ROA0.0515**0.0726***0.0396*0.00650.00630.0062(246)(2.73)(1.89)(1.09)(1.04)(1.04)CEO_DUALITY0.01420.00500.01300.00120.0012(1.35)(1.37)(1.53)(0.79)(0.79)(1.55)LNCEO_TENURE-0.1874***-0.021-0.0021-0.0020-0.00211-16.45)(-15.17)(-15.7)(-1.55)(-1.45)(-1.45)LNBOARD_SIZE-0.0042-0.1562***-0.00340.0014-0.0025-0.00251.710(1.72)(1.48)(-0.64)(-0.47)(-0.47)LNANLYSTS0.0227***0.0285**0.0242**-0.0028**-0.0028**0.021(1.71)(1.72)(1.48)(-0.64)(-0.47)(-0.47)LNANLYSTS0.0227***0.0285**0.0242***-0.0028**-0.0028***0.021(3.79)(4.28)0.0210.0120.012RELATIVESIZE0.00230.0023ALLSTOCK0.00230.0023ALLSTOCK0.0012(-1.69)RELATIPDBID0.0024(-0.19)TENDEROFFER(0.25)(0.34)TENDEROFFER(0.051-0.0056*COMPETINGBID(0.012COMPETINGEID1.600(3.04) <td>TOBINQ</td> <td>-0.0034</td> <td>-0.0029</td> <td>-0.0031</td> <td>0.0028***</td> <td>0.0028***</td> <td>0.0028***</td> <td></td>  | TOBINQ                   | -0.0034            | -0.0029     | -0.0031     | 0.0028***         | 0.0028***        | 0.0028***  |                       |
| CEO_DUALITY         0.0142         0.0065         0.0136         0.0013         0.0012         0.0012           LNCEO_TENURE         (-135)         (0.54)         (1.30)         (0.82)         (0.79)         (0.79)           LNCEO_TENURE         -0.1874***         -0.0121         -0.0020         -0.0020         -0.0021           LNBOARD_SIZE         -0.0024         -0.1562***         -0.0038         0.0011         -0.0001         -0.0021           LNBOARD_SIZE         -0.022         (-6.62)         (-0.21)         (0.40)         (-0.04)         (-0.04)           BOARD_INDEP         0.0540*         0.0585**         0.044*         -0.0028**         -0.0028**         -0.0028**         -0.0028***         -0.0128***                                    | ROA                      | 0.0515**           | 0.0726***   | 0.0396*     | 0.0065            | 0.0063           | 0.0062     |                       |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$   | CEO_DUALITY              | 0.0142             | 0.0065      | 0.0136      | 0.0013            | 0.0012           | 0.0012     |                       |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$   | LNCEO_TENURE             | -0.1874***         | -0.1919***  | -0.2127***  | -0.0021           | -0.0020          | -0.0020    |                       |
| BOARD_INDEP         0.0540*         0.0588*         0.0444         -0.0034         -0.0025         -0.0025           I.71)         (1.72)         (1.48)         (-0.64)         (-0.47)         (-0.47)           LNANALYSTS         0.0227***         0.0285***         0.0242***         -0.0028***         -0.0028***         -0.0028***           RELATIVESIZE         -         0.0227***         0.0242***         -0.0020         0.0012         0.0012           RELATIVESIZE         -         -         0.0020         0.0012         0.0012           ALLCASH         -         -         0.0023         0.0023         0.0023           ALLSTOCK         -         -         -         -         0.0115***         -0.0120***           RELATEDBID         -         -         -         -         -0.0115***         -0.0120***         -0.0120***           RELATEDBID         -         -         -         -         -         0.0004         0.0005         0.0005           RELATEDBID         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -   | LNBOARD_SIZE             | -0.0042            | -0.1562***  | -0.0038     | 0.0001            | -0.0001          | -0.0001    |                       |
| $ \begin{array}{llllllllllllllllllllllllllllllllllll$   | BOARD_INDEP              | 0.0540*            | 0.0588*     | 0.0444      | -0.0034           | -0.0025          | -0.0025    |                       |
| RELATIVESIZE0.00200.00120.00120.0012ALLCASH0.00230.00230.0023ALLSTOCK(1.62)(1.59)(1.59)ALLSTOCK0.0115***-0.0120***RELATEDBID(0.25)(0.34)(0.34)TENDEROFFER0.0057**-0.0056**-0.0056**COMPETINGBID0.0051(0.34)CONSTANT0.06350.4850***0.3755***0.01260.01320.0132(0.60)(3.04)(3.74)(0.90)(0.95)(0.94)Year fixed effectsYesYesYesYesYesYesR <sup>2</sup> 0.52500.46670.54280.03840.03850.0385N8,9398,9398,9398,9398,9398,9398,939  | LNANALYSTS               | 0.0227***          | 0.0285***   | 0.0242***   | -0.0028***        | -0.0028***       | -0.0028*** |                       |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$   | RELATIVESIZE             | (3.79)<br>-        | (4.28)<br>- | (4.20)<br>- | 0.0020            | 0.0012           | 0.0012     |                       |
| ALLSTOCK $-0.0120^{***}$ $-0.0120^{***}$ $-0.0120^{***}$ RELATEDBID0.00040.00050.0005RELATEDBID0.00040.00050.0005TENDEROFFER0.0057**-0.0056**COMPETINGBID0.0051COMPETINGBID0.0051CONSTANT0.06350.4850***0.3755***0.01260.0132(0.60)(3.04)(3.74)(0.90)(0.95)(0.94)Year fixed effectsYesYesYesYesYesPricesR <sup>2</sup> 0.52500.46670.54280.03840.03850.0385N8,9398,9398,9398,9398,9398,939Weak identification   | ALLCASH                  | _                  | _           | _           |                   | 0.0023           | 0.0023     |                       |
| RELATEDBID $   0.0004$ $0.0005$ $0.0005$ $0.0005$ TENDEROFFER $  0.0057**$ $(0.34)$ $(0.34)$ TENDEROFFER $  -0.0057**$ $-0.0056**$ $-0.0056**$ COMPETINGBID $   -0.0051$ $-0.0046$ $-0.0048$ CONSTANT $0.0635$ $0.4850***$ $0.3755***$ $0.0126$ $0.0132$ $0.0132$ CONSTANT $0.06035$ $0.4850***$ $0.3755***$ $0.0126$ $0.0132$ $0.0132$ Vear fixed effectsYesYesYesYesYesIndustry fixedYesYesYesYesYes $R^2$ $0.5250$ $0.4667$ $0.5428$ $0.0384$ $0.0385$ $0.0385$ N $8,939$ $8,939$ $8,939$ $8,939$ $8,939$ $8,939$  | ALLSTOCK                 | _                  | -           | -           | -0.0115***        | -0.0120***       | -0.0120*** |                       |
| TENDEROFFER $-0.0057^{**}$ $-0.0056^{**}$ $-0.0056^{**}$ COMPETINGBID $(-2.18)$ $(-2.18)$ CONSTANT0.06350.4850^{**}0.3755^{***}0.01260.01320.0132(0.60)(3.04)(3.74)(0.90)(0.95)(0.94)Year fixed effectsYesYesYesYesYesIndustry fixedYesYesYesYesYes $R^2$ 0.52500.46670.54280.03840.03850.0385N8,9398,9398,9398,9398,9398,939Weak identification  | RELATEDBID               | _                  | -           | -           | 0.0004            | 0.0005           | 0.0005     |                       |
| $\begin{array}{cccccccccccccccccccccccccccccccccccc$  | TENDEROFFER              | -                  | _           | -           | -0.0057**         | -0.0056**        | -0.0056**  |                       |
| $\begin{array}{cccccccccccccccccccccccccccccccccccc$  | COMPETINGBID             | _                  | _           | -           | -0.0051           | -0.0046          | -0.0048    |                       |
| Year fixed effectsYesYesYesYesYesYesIndustry fixedYesYesYesYesYesYeseffects $R^2$ 0.52500.46670.54280.03840.03850.0385N8,9398,9398,9398,9398,9398,939Weak identification  | CONSTANT                 |                    |             |             | 0.0126            | 0.0132           | 0.0132     |                       |
| $ \begin{array}{llllllllllllllllllllllllllllllllllll$   | Voor fixed offects       |                    |             |             |                   |                  |            |                       |
| R <sup>2</sup> 0.5250         0.4667         0.5428         0.0384         0.0385         0.0385           N         8,939         8,939         8,939         8,939         8,939         8,939         8,939           Weak identification  | Industry fixed           |                    |             |             |                   |                  |            |                       |
| N 8,939 8,939 8,939 8,939 8,939 8,939 8,939 Weak identification   |                          | 0.5250             | 0.4667      | 0.5428      | 0.0384            | 0.0385           | 0.0385     |                       |
| test  | N<br>Weak identification |                    |             |             |                   |                  |            |                       |
| (continued)   | test                     |                    |             |             |                   |                  | ( h        |                       |

#### Table 6. Continued

|  | Panel A: firs<br>(1)<br>PCO_OPT | t stage<br>(2)<br>NCO_OPT | (3)<br>RCO_OPT | Panel B: sec<br>(4)<br>CAR | ond stage<br>(5)<br>CAR | (6)<br>CAR      |
|--|---------------------------------|---------------------------|----------------|----------------------------|-------------------------|-----------------|
| Kleibergen-Paap rk<br>Wald F statistic | 3682.84                         | 3890.09                   | 3419.81        | _                          | _                       | -               |
| <i>p</i> -value                        | 0.0000                          | 0.0000                    | 0.0000         |                            |                         |                 |
| Test of endogeneity:                   | Durbin–Wu–H                     | ausman test               |                |                            |                         |                 |
| <i>F</i> statistics <i>p</i> -value    | -                               | _                         | _              | 11.47<br>0.0007            | 12.28<br>0.0005         | 11.58<br>0.0007 |
| Note(s). This table r                  | oports the opd                  | ogonoity corroc           | tod rogrossion | roculte by om              | lowing the 28           | IS rogrossions  |

**Note(s):** This table reports the endogeneity-corrected regression results by employing the 2SLS regressions approach. Panel A reports first-stage regression output where the categorical PCO\_OPT, NCO\_OPT and RCO\_OPT variables are regressed on the respective instrumental variables with other firm-specific control variables. Panel B reports the second-stage regression output where the acquisition performance is regressed on the instrumented PCO\_OPT, NCO\_OPT and RCO\_OPT variables and other control variables. Robust two-tailed *t*-statistics clustered by firm are presented in parentheses. The superscripts <sup>\*\*\*</sup>, <sup>\*\*\*</sup> and <sup>\*</sup> correspond to statistical significance at the 1%, 5% and 10% levels, respectively **Source(s):** Authors' own work

the inference model to estimate effects for variables of interest (Belloni, Chernozhukov, & Hansen, 2014). We present the results of the double-selection Lasso method in Panel B of Table 7, where the control variables are selected by the Lasso model for the variables of interest to be included in the model. One should note that double selection Lasso does not provide estimates of the coefficients on the control variables or their standard errors (Belloni *et al.*, 2014). However, the estimation results for three variables of interest representing board cooption are similar to baseline regression results.

#### 4.3 Additional analysis

4.3.1 Role of CEO characteristics. We have evidenced that board co-option negatively affects returns to acquirers surrounding the acquisition announcement. However, prior studies suggest that CEO characteristics such as CEO age, risk-taking incentives, tenure and gender can significantly influence acquisition decisions and outcomes. Younger CEOs face stronger incentives to pursue acquisitions since they have longer career horizons over which to reap the benefits. While compensation benefits associated with empire-building suggest CEOs have greater incentives to pursue acquisitions earlier in their career, career concerns may make vounger CEOs reluctant to jeopardise future earnings and therefore avoid risky activities (Yim, 2013). Furthermore, the organisational legitimacy literature shows that the pressure for legitimacy should be greater for the CEOs who are younger who want to establish legitimacy also for a career perspective (De Franco, Hou, & Ma, 2022). Similarly, CEO vega, a measure of risk-taking incentives, captures the extent to which a CEO's compensation is tied to risk, such as stock options. CEOs with higher vega are more likely to pursue riskier acquisitions due to the potential for greater personal financial gain (Hagendorff & Vallascas, 2011). CEO tenure plays a critical role as well, as long-tenured CEOs may have established stronger ties with the board, giving co-opted directors greater influence over strategic decisions such as acquisitions. Finally, CEO gender is another important dimension to consider. Prior research has shown that female CEOs may approach acquisitions with different leadership styles or risk profiles compared to their male counterparts (Cumming, Leung, & Rui, 2015; Luong et al., 2023). The dynamics between co-opted boards and female CEOs could vary significantly, potentially leading to different acquisition outcomes (Frye & Pham, 2018). These arguments indicate a differential relationship between board co-options and acquisition performance across the CEO characteristic divide. To account for these possible influences, we created four variables:

|                          | (1)<br>CAR      | (2)<br>CAR      | (3)<br>CAR      | (4)<br>CAR      | (5)<br>CAR      | (6)<br>CAR      | (7)<br>CAR | (8)<br>CAR | (9)<br>CAR |
|--------------------------|-----------------|-----------------|-----------------|-----------------|-----------------|-----------------|------------|------------|------------|
| Panel A: Lasso selection | n models        |                 |                 |                 |                 |                 |            |            |            |
| PCO_OPT                  | -0.0042         | -0.0046         | -0.0005         | -               | _               | -               | -          | -          | _          |
| NCO_OPT                  | _               | -               | -               | -0.0004         | -0.0005         | -0.0002         | -          | -          | _          |
| RCO_OPT                  | -               | -               | _               | -               | -               | -               | -0.0046    | -0.0048    | _          |
| SIZE                     | -0.0016         | -0.0020         | -0.0007         | -0.0014         | -0.0015         | -0.0006         | -0.0016    | -0.0019    | -0.0007    |
| LEV                      | 0.0116          | 0.0135          | 0.0013          | 0.0116          | 0.0137          | 0.0009          | 0.0120     | 0.0135     | 0.0013     |
| CASH                     | -0.0104         | -0.0147         | Х               | -0.0113         | -0.0143         | Х               | -0.0113    | -0.0147    | Х          |
| TOBINQ                   | 0.0016          | 0.0022          | Х               | 0.0018          | 0.0024          | Х               | 0.0018     | 0.0023     | Х          |
| ROA                      | 0.0039          | 0.0038          | Х               | 0.0043          | 0.0057          | Х               | 0.0047     | 0.0045     | Х          |
| CEO_DUALITY              | -0.0001         | Х               | Х               | Х               | Х               | Х               | Х          | Х          | Х          |
| LNCEO_TENURE             | Х               | Х               | Х               | -0.0005         | -0.0016         | Х               | -0.0001    | Х          | Х          |
| LNBOARD_SIZE             | Х               | Х               | Х               | Х               | Х               | Х               | Х          | Х          | Х          |
| BOARD_INDEP              | Х               | Х               | Х               | Х               | Х               | Х               | Х          | Х          | Х          |
| LNANALYSTS               | -0.0020         | -0.0021         | -0.0006         | -0.0021         | -0.0025         | -0.0006         | -0.0022    | -0.0022    | -0.0006    |
| RELATIVESIZE             | 0.0006          | Х               | Х               | 0.0008          | 0.0008          | Х               | 0.0009     | Х          | Х          |
| ALLCASH                  | 0.0011          | 0.0005          | Х               | 0.0013          | 0.0019          | Х               | 0.0014     | 0.0009     | Х          |
| ALLSTOCK                 | -0.0111         | -0.0126         | -0.0048         | -0.0111         | -0.0121         | -0.0049         | -0.0112    | -0.0125    | -0.0048    |
| RELATEDBID               | -0.0003         | Х               | Х               | -0.0004         | -0.0004         | Х               | -0.0006    | Х          | Х          |
| TENDEROFFER              | -0.0034         | -0.0033         | Х               | -0.0038         | -0.0050         | Х               | -0.0040    | -0.0038    | Х          |
| COMPETINGBID             | -0.0011         | Х               | Х               | -0.0018         | -0.0021         | Х               | -0.0020    | Х          | Х          |
| Year fixed effects       | Yes             | Yes             | Yes             | Yes             | Yes             | Yes             | Yes        | Yes        | Yes        |
| Industry fixed effects   | Yes             | Yes             | Yes             | Yes             | Yes             | Yes             | Yes        | Yes        | Yes        |
| Ν                        | 8,939           | 8,939           | 8,939           | 8,939           | 8,939           | 8,939           | 8,939      | 8,939      | 8,939      |
| Panel B: Lasso inferenc  | e models        |                 |                 |                 |                 |                 |            |            |            |
| PCO_OPT                  | $-0.0071^{***}$ | $-0.0071^{***}$ | $-0.0074^{***}$ | -               | -               | -               | -          | -          | -          |
|                          | (0.00)          | (0.00)          | (0.00)          |                 |                 |                 |            |            |            |
| NCO_OPT                  | _               | _               | _               | $-0.0007^{***}$ | $-0.0007^{***}$ | $-0.0007^{***}$ | -          | -          | _          |
|                          |                 |                 |                 | (0.00)          | (0.00)          | (0.00)          |            |            |            |

## Table 7. Lasso selection and Lasso reference models

(continued)

#### Table 7. Continued

|                                 | (1)<br>CAR   | (2)<br>CAR   | (3)<br>CAR   | (4)<br>CAR   | (5)<br>CAR   | (6)<br>CAR   | (7)<br>CAR            | (8)<br>CAR            | (9)<br>CAR             |
|---------------------------------|--------------|--------------|--------------|--------------|--------------|--------------|-----------------------|-----------------------|------------------------|
| RCO_OPT                         | -            | -            | -            | -            | -            | -            | $-0.0064^{**}$ (0.00) | $-0.0064^{**}$ (0.00) | $-0.0068^{***}$ (0.00) |
| Controls                        | Yes                   | Yes                   | Yes                    |
| Year fixed effects              | Yes                   | Yes                   | Yes                    |
| Industry fixed effects <i>N</i> | Yes<br>8,939          | Yes<br>8,939          | Yes<br>8,939           |

**Note(s):** This table presents the results of the Lasso selection model for board co-option and acquisition performance. The Lasso model selection method (Panel A) is used to estimate effects for potential independent and control variables to be included in the model. Model 1 uses adaptive Lasso selection model; Model 2 is Lasso selection model with cross-validation method; Model 3 employs plug-in method. Omitted variables by the Lasso selection are denoted as (X). Panel B presents Lasso inference models based on double-selection Lasso regression

**Source(s):** Authors' own work

the first for young CEOs, taking the value of one if the CEOs are younger than 65 years old; the China Accounting second for CEO vega, differentiating high and low risk-taking incentives; the third for CEO tenure, categorising short and long tenure; and the fourth for CEO gender, distinguishing between male and female CEOs We partition the sample based on these CEO characteristics and re-estimate equation (1) for each sub-sample. The results are reported in Table 8.

In Panel A, we find that the coefficients generated by board co-option variables are negative and significant in the old CEO subsample, aligning with the acquisition literature that old CEOs are less incentivised in pursuing acquisitions. By contrast, board co-option variables generate marginal significant or insignificant coefficients in the young CEO subsample, suggesting that the negative impact of board co-option on acquisition performance is more pronounced if CEOs are young. The chi-square tests indicate that the differences in the estimations are statistically different. It appears that, being incentivised by long-term career perspective and under pressure for legitimacy, younger CEOs tend to make the board co-option and acquisition performance less pronounced. In Panel B, our results indicate that co-option has a stronger negative effect on acquisition returns for high vega CEOs, reflecting that risktaking CEOs may not be properly monitored when boards are dominated by co-opted directors. For low vega CEOs, the impact of co-option is less pronounced, likely reflecting their more conservative approach to acquisitions. Panel C examines the role of CEO tenure. We find that the negative impact of board co-option is stronger for long-tenured CEOs, who may have built stronger ties with the board thus exhibiting higher influence in acquisition decisions. In contrast, for short-tenured CEOs, the effects of co-option are weaker, as these CEOs may not have established as much control over the board, resulting in less entrenched dynamics. Panel D reports that co-opted boards have a more significant negative impact on acquisition performance for male CEOs compared to female CEOs. For male CEOs, co-opted boards exert stronger control over decision-making, leading to more negative acquisition outcomes.

4.3.2 Internal and external monitorina. 4.3.2.1 Entrenchment index. Thus far, we have revealed that the presence of board co-option reduces announcement returns to acquirers. This effect, however, presumably varies with differences in the level of internal control. This is because managers of firms with weak corporate governance could make strategic decisions to reduce stakeholder power which affects corporate efficiency negatively (Hill & Jones, 1992). The purposive managerial actions include withholding non-public or adverse financial information to cover up their value-destroying actions. Moreover, weak internal control provides opportunities for managers to pursue short-term investments which may be costly to shareholders' wealth (Armstrong, Balakrishnan, & Cohen, 2012; Ulupinar, 2018).

Following prior studies (Bebchuk, Cohen and Wang, 2013), we use the entrenchment index (EINDEX) to divide companies into two groups as weakly governed firms and well-governed firms: the yearly median *EINDEX* is used as the cut-off point. Accordingly, *HIGH\_EINDEX* takes the value of one if the firm's *EINDEX* is greater than or equal to the yearly median EINDEX, and zero otherwise. We re-estimate our baseline model in the two subsamples. This provides a powerful test because internal control can significantly impact acquisition decisions and the potential co-opted relationship between the CEO and directors (Masulis *et al.*, 2007). Panel A of Table 9 reports that the coefficients of board co-option proxies in Columns (1), (3) and (5) for firms with a high EINDEX, representing weak internal control, are negative and significant at the 5% level or higher (coefficient = -0.0106, -0.0010 and -0.0102, respectively, *p*-value <0.1), while being insignificant in Column (2), (4) and (6) for the low-*EINDEX* subsample. These results suggest that the relationship between board co-option and acquisition performance is more severe in poorly governed firms. This finding suggests that firms with entrenched boards are less likely to actively monitor the acquisition decision of the highly co-opted board, leading them to drive the negative relationship between board cooption and acquisition in our sample.

and Finance Review

Panel A: CEO age Young Young YoungCEO Old CEO CEO OldCEO CEO Old CEO (1)(2)(3) (4) (5) (6) CÁR CÁR CÁR CÁR CÁR CÁR -0.0059\* -0.0083\*\*\* PCO\_OPT (-1.74)(-3.20)-0.0006\*-0.0007\*\*\* NCO\_OPT (-1.83)(-3.19)RCO\_OPT -0.0053-0.0075\*\*\*(-2.92)(-1.55)0.0303 -0.0117CONSTANT 0.0241 -0.01740.0267 -0.0164(0.91)(1.00)(-0.78)(1.14)(-0.56)(-0.82)(0.07)(0.00)Chi-square (0.06)difference Controls Yes Yes Yes Yes Yes Yes Year fixed effects Yes Yes Yes Yes Yes Yes Industry fixed Yes Yes Yes Yes Yes Yes effects  $R^2$ 0.0481 0.0529 0.0481 0.0526 0.0480 0.0526 Ν 3,849 5,090 3,849 3,849 5,090 5,090 Panel B: CEO vega High CEO Low CEO High CEO Low CEO High CEO Low CEO vega vega vega vega vega vega (1) CAR (2) CAR (3)(4) (5) (6) ĊÁR **CÁR ÒÁR CÁR** PCO\_OPT -0.0119\*\*\* -0.0042\* (-3.13)(-1.73)-0.0009\*\*\* -0.0004\*NCO\_OPT (-2.76)(-1.95)RCO\_OPT -0.0113\*\*\*-0.0036(-2.97)(-1.49)CONSTANT 0.0051 0.0122 -0.00770.0084 0.0027 0.0099 (0.31)(0.74)(-0.46)(0.51)(-0.16)(0.60)Chi-square (0.05)(0.07)(0.00)difference Controls Yes Yes Yes Yes Yes Yes Year fixed Yes Yes Yes Yes Yes Yes effects Industry fixed Yes Yes Yes Yes Yes Yes effects  $R^2$ 0.0909 0.0413 0.0892 0.0414 0.0904 0.0412 Ν 2,222 6,717 2,222 6,717 2,222 6,717 Panel C: CEO tenure Short CEO Short CEO Long CEO Long CEO Short CEO Long CEO tenure tenure tenure tenure tenure tenure (1)(3)(4) (5) (6) (2) CAR CAR CAR CAR CAR CAR PCO\_OPT -0.0035-0.0093\*\*\* (-1.42)(-2.84)(continued)

Table 8. Co-opted board and market reactions to acquisition announcements: the role of CEO characteristics

| Table 8. C | ontinued |
|------------|----------|
|------------|----------|

China Accounting and Finance

| Panel C: CEO ter                  | nure                              |                                  |                                   |                                  |                                      |                                  | Review |
|-----------------------------------|-----------------------------------|----------------------------------|-----------------------------------|----------------------------------|--------------------------------------|----------------------------------|--------|
|                                   | Short CEO<br>tenure<br>(1)<br>CAR | Long CEO<br>tenure<br>(2)<br>CAR | Short CEO<br>tenure<br>(3)<br>CAR | Long CEO<br>tenure<br>(4)<br>CAR | Short CEO<br>tenure<br>(5)<br>CAR    | Long CEO<br>tenure<br>(6)<br>CAR |        |
| NCO_OPT                           |                                   |                                  | -0.0004 $(-1.58)$                 | $-0.0007^{**}$<br>(-2.58)        |                                      |                                  |        |
| RCO_OPT                           |                                   |                                  | ( 1.50)                           | ( 2.50)                          | -0.0046*<br>(-1.82)                  | $-0.0087^{***}$<br>(-2.72)       |        |
| CONSTANT                          | 0.0323*<br>(1.79)<br>(0.00)       | -0.0832*<br>(-1.83)              | 0.0287<br>(1.56)<br>(0.00)        | -0.0893*<br>(-1.91)              | (1.62)<br>0.0297<br>(1.60)<br>(0.06) | (-2.72)<br>-0.0888**<br>(-2.20)  |        |
| Chi-square<br>difference          | . ,                               |                                  | . ,                               |                                  |                                      |                                  |        |
| Controls<br>Year fixed<br>effects | Yes<br>Yes                        | Yes<br>Yes                       | Yes<br>Yes                        | Yes<br>Yes                       | Yes<br>Yes                           | Yes<br>Yes                       |        |
| Industry fixed<br>effects         | Yes                               | Yes                              | Yes                               | Yes                              | Yes                                  | Yes                              |        |
| R <sup>2</sup><br>N               | 0.0434<br>5,693                   | 0.0608<br>3,246                  | 0.0434<br>5,693                   | 0.0600<br>3,246                  | 0.0435<br>5,693                      | 0.0604<br>3,246                  |        |
| Panel D: CEO ge                   | nder                              |                                  |                                   |                                  |                                      |                                  |        |
|                                   | Female<br>CEO<br>(1)<br>CAR       | Male CEO<br>(2)<br>CAR           | Female<br>CEO<br>(3)<br>CAR       | Male CEO<br>(4)<br>CAR           | Female<br>CEO<br>(5)<br>CAR          | Male CEO<br>(6)<br>CAR           |        |
| PCO_OPT                           | -0.0035 $(-0.58)$                 |                                  | *                                 |                                  |                                      |                                  |        |
| NCO_OPT                           | (-0.50)                           | (-2.23)                          | -0.0003 $(-0.63)$                 | $-0.0004^{**}$<br>(-1.98)        | k                                    |                                  |        |
| RCO_OPT                           |                                   |                                  | ()                                | ( 1.00)                          | -0.0045<br>(-0.65)                   | $-0.0056^{**}$<br>(-2.25)        |        |
| CONSTANT                          | 0.0154<br>(0.48)                  | 0.0770<br>(1.60)                 | 0.0115<br>(0.35)                  | 0.0735<br>(1.54)                 | 0.0128<br>(0.38)                     | 0.0732<br>(1.53)                 |        |
| Chi-square<br>difference          | (0.00)                            | × /                              | (0.00)                            | · · ·                            | (0.00)                               | ` ´                              |        |
| Controls                          | Yes                               | Yes                              | Yes                               | Yes                              | Yes                                  | Yes                              |        |
| Year fixed effects                | s Yes                             | Yes                              | Yes                               | Yes                              | Yes                                  | Yes                              |        |
| Industry fixed<br>effects         | Yes                               | Yes                              | Yes                               | Yes                              | Yes                                  | Yes                              |        |
| $R^2$                             | 0.1373                            | 0.0461                           | 0.1373                            | 0.0458                           | 0.1374                               | 0.0461                           |        |
| Ν                                 | 903                               | 5,434                            | 903                               | 5,434                            | 903                                  | 5,434                            |        |
| Note(s): This tab                 | le presents the                   | regression result                | ts of the effect                  | of co-opted boa                  | rd on acquisit                       | ion performance                  |        |

**Note(s):** This table presents the regression results of the effect of co-opted board on acquisition performance with other control variables for young versus old CEO in Panel A, low versus high CEO vega in Panel B, short versus long CEO tenure in Panel C and female versus male CEO in Panel D. Robust two-tailed *t*-statistics clustered by firm are presented in parentheses. The superscripts **\*\*\***, **\*\*** and **\*** correspond to statistical significance at the 1%, 5% and 10% levels, respectively. We present the variable definitions in the Appendix **Source(s):** Authors' own work

4.3.2.2 Institutional ownership. A potentially confounding factor related to our investigation of the relation between board co-option and announcement returns to acquirers is that firms with high institutional ownership are assumed to face more exposure from the public than their counterparts. It has been established that the increased scrutiny of the relationship between

CAFR

**Table 9.** Co-opted board and market reactions to acquisition announcements: the role of internal and external monitoring

| Panel A: entrenchmen   | t index           |                            |            |            |                   |                   |
|------------------------|-------------------|----------------------------|------------|------------|-------------------|-------------------|
|                        | High              | Low                        | High       | Low        | High              | Low               |
|                        | eindex            | eindex                     | eindex     | eindex     | eindex            | eindex            |
|                        | (1)               | (2)                        | (3)        | (4)        | (5)               | (6)               |
|                        | CAR               | CAR                        | CAR        | CAR        | CAR               | CAR               |
|                        | -                 | Grift                      | Grift      | Critt      | Grift             | Grift             |
| PCO_OPT                | -0.0106*          | -0.0019                    | -          | -          | -                 | -                 |
|                        | (-1.90)           | (-0.54)                    |            |            |                   |                   |
| NCO_OPT                | _                 | _                          | -0.0010*   | -0.0003    | _                 | -                 |
|                        |                   |                            | (-1.91)    | (-0.83)    |                   |                   |
| RCO_OPT                | _                 | _                          | _          | _          | -0.0102*          | -0.0016           |
| -                      |                   |                            |            |            | (-1.83)           | (-0.47)           |
| CONSTANT               | 0.0234            | 0.0169                     | 0.0130     | 0.0146     | 0.0165            | 0.0159            |
| 0011011111             | (0.92)            | (1.13)                     | (0.49)     | (0.96)     | (0.63)            | (1.06)            |
| Controls               | Yes               | Yes                        | Yes        | Yes        | Yes               | Yes               |
| Year fixed effects     | Yes               | Yes                        | Yes        | Yes        | Yes               | Yes               |
| Industry fixed         | Yes               | Yes                        | Yes        | Yes        | Yes               | Yes               |
| effects                | 165               | 165                        | 165        | 165        | 165               | 165               |
| $R^2$                  | 0.1138            | 0.0612                     | 0.1135     | 0.0613     | 0.1136            | 0.0612            |
|                        |                   |                            |            |            |                   |                   |
| Ν                      | 954               | 2,763                      | 954        | 2,763      | 954               | 2,763             |
| Panel B: Institutional |                   | <b>T T</b> .               | TT' J ' .  | <b>T</b>   | <b>TT</b> 1       | <b>.</b>          |
|                        | High insto        | Low insto                  | High insto | Low insto  | High insto        | Low insto         |
|                        | (1)               | (2)                        | (3)        | (4)        | (5)               | (6)               |
|                        | CAR               | CAR                        | CAR        | CAR        | CAR               | CAR               |
| PCO_OPT                | -0.0028 $(-0.97)$ | $-0.0107^{***}$<br>(-3.52) | _          | -          | -                 | -                 |
| NCO_OPT                | _                 | _                          | -0.0003    | -0.0010*** | _                 | _                 |
| 1100_011               |                   |                            | (-1.11)    | (-3.54)    |                   |                   |
| RCO OPT                |                   |                            | ( 1.11)    | ( 3.34)    | -0.0023           | $-0.0099^{**}$    |
| ICO_OF1                | -                 | -                          | _          | -          | (-0.80)           | (-3.27)           |
| CONSTANT               | 0.0295            | 0.0149                     | 0.0268     | 0.0063     | (-0.80)<br>0.0281 | (-3.27)<br>0.0084 |
| CONSTANT               |                   |                            |            |            |                   |                   |
|                        | (0.78)            | (1.18)                     | (0.71)     | (0.49)     | (0.74)            | (0.66)            |
| Controls               | Yes               | Yes                        | Yes        | Yes        | Yes               | Yes               |
| Year fixed effects     | Yes               | Yes                        | Yes        | Yes        | Yes               | Yes               |
| Industry fixed effects | Yes               | Yes                        | Yes        | Yes        | Yes               | Yes               |
| $R^2$                  | 0.0525            | 0.0426                     | 0.0525     | 0.0422     | 0.0524            | 0.0423            |
| Ν                      | 4,284             | 4,655                      | 4,284      | 4,655      | 4,284             | 4,655             |
|                        |                   |                            |            |            |                   |                   |

**Note(s):** This table presents the regression results of the effect of co-opted board on acquisition performance with other control variables for high versus low entrenchment index in Panel A and high versus low institutional shareholdings in Panel B. Robust two-tailed *t*-statistics clustered by the firm are presented in parentheses. The superscripts \*\*\*, \*\* and \* correspond to statistical significance at the 1%, 5% and 10% levels, respectively. We present the variable definitions in the Appendix

Source(s): Authors' own work

board members has fallen more heavily on those firms with large institutional shares (Edmans & Holderness, 2017). Arguably, managers in those firms tend to be more reluctant to exercise self-interest in corporate actions. Therefore, we anticipate that the negative impact of board cooption is less pronounced in this group in response to such exposure to scrutiny. To empirically test our prediction, we split our sample into two groups: (1) firms with high institutional ownership and (2) firms with low institutional ownership. We estimate equation (1) separately for these two groups. Panel B of Table 9 reports the results. We find that the coefficients of board co-option on acquisition performance only remain negative and significant in the low-institutional-ownership subsample while becoming insignificant in the high-institutionalownership subsample, meaning that the co-opted board and acquisition performance relationship diminishes when firms are under the scrutiny of institutional shareholders. Therefore, we conclude that institutional ownership alleviates the negative impact of board cooption on acquisition decisions.

4.3.3 Co-opted board and acquisition performance: the role of target board co-option. Next, we empirically examine the role of target firms' co-option boards on the association between bidder co-option board and acquisition performance. We argue that the negative association between bidder co-option and acquisition performance is more pronounced when target firms also have co-opted boards. We present the results in Table 10. Consistent with our predictions, the results show that the negative impact of board co-option on acquisition performance is more evident when the target board's co-option is higher. The findings remain consistent in all three measures of board co-option reported in Panel A of Table 10. However, we do not find the negative and significant impact of bidder co-opted board and market reactions when target firms have a lower proportion of co-opted directors reported in Panel B of Table 10. The findings imply that investors evaluate whether the acquisition decision further deteriorates the co-option structure of the bidder's board following the acquisition of targets with highly co-opted boards.

4.3.4 Deal characteristics. Prior studies find that the market response to acquisition announcements depends on the target's listing status (Chang, 1998; Officer, 2007), the method of payment used (Travlos, 1987), the industry in which the acquirer and target operate (Morck *et al.*, 1990; Shleifer & Vishny, 2003), the size of the deal (Alexandridis, Fuller, Terhaar, & Travlos, 2013) and the location of the target (Bertrand & Betschinger, 2012; Danbolt & Maciver, 2012). We therefore partition the sample according to these characteristics and estimate equation (1) separately for these groups. The findings are reported in Table 11.

In Panel A, when we split the sample into public and non-public targets, we find that the negative association between board co-option and announcement returns is mainly driven by the acquisition of public targets. In Panel B, the negative impact of the presence of co-opted

|                        | Panel A: high co-opted target board |                           |                           | Panel B: low co-opted target boa |                    |                    |
|------------------------|-------------------------------------|---------------------------|---------------------------|----------------------------------|--------------------|--------------------|
|                        | (1)                                 | (2)                       | (3)                       | (4)                              | (5)                | (6)                |
|                        | CAR                                 | CAR                       | CAR                       | CAR                              | CAR                | CAR                |
| PCO_OPT                | -0.0079**<br>(-2.36)                | -                         | -                         | -0.0035 $(-1.10)$                | -                  | -                  |
| NCO_OPT                | _                                   | $-0.0008^{**}$<br>(-2.53) | -                         | _                                | -0.0004<br>(-1.18) | -                  |
| RCO_OPT                | -                                   | _                         | $-0.0075^{**}$<br>(-2.26) | _                                | _                  | -0.0027<br>(-0.85) |
| CONSTANT               | -0.0227<br>(-1.32)                  | -0.0296*<br>(-1.66)       | -0.0277<br>(-1.57)        | -0.0118<br>(-0.43)               | -0.0152<br>(-0.55) | -0.0138<br>(-0.50) |
| Controls               | Yes                                 | Yes                       | Yes                       | Yes                              | Yes                | Yes                |
| Year fixed effects     | Yes                                 | Yes                       | Yes                       | Yes                              | Yes                | Yes                |
| Industry fixed effects | Yes                                 | Yes                       | Yes                       | Yes                              | Yes                | Yes                |
| $R^2$                  | 0.0424                              | 0.0424                    | 0.0423                    | 0.0592                           | 0.0592             | 0.0591             |
| Ν                      | 3,248                               | 3,248                     | 3,248                     | 3,567                            | 3,567              | 3,567              |

 Table 10.
 Co-opted board and market reactions to acquisition announcements: the role of target board co-option

**Note(s):** This table presents the regression results of the effect of co-opted board on acquisition performance for high and low co-opted target board in Panel A and Panel B, respectively. Robust two-tailed *t*-statistics clustered by firm are presented in parentheses. The superscripts \*\*\*, \*\* and \* correspond to statistical significance at the 1%, 5% and 10% levels, respectively. We present the variable definitions in the Appendix **Source(s):** Authors' own work

China Accounting and Finance Review

Table 11. Co-opted board and market reactions to acquisition announcements: the role of deal characteristics

| Panel A: public an        | d private deals            | Duir                          |                            | Duizzete                      |                            | Duiz                          |
|---------------------------|----------------------------|-------------------------------|----------------------------|-------------------------------|----------------------------|-------------------------------|
|                           | Public deal<br>(1)<br>CAR  | Private<br>deal<br>(2)<br>CAR | Public deal<br>(3)<br>CAR  | Private<br>deal<br>(4)<br>CAR | Public deal<br>(5)<br>CAR  | Private<br>deal<br>(6)<br>CAR |
| PCO_OPT                   | $-0.0080^{***}$<br>(-2.99) | -0.0053<br>(-1.63)            | -                          | -                             | _                          | -                             |
| NCO_OPT                   | -                          | -                             | $-0.0007^{***}$<br>(-3.08) | $-0.0005^{*}$<br>(-1.68)      | -                          | -                             |
| RCO_OPT                   | _                          | -                             | -                          | -                             | $-0.0073^{***}$<br>(-2.76) | -0.0047<br>(-1.44)            |
| CONSTANT                  | 0.0305<br>(1.34)           | -0.0195<br>(-0.74)            | 0.0235<br>(1.05)           | -0.0236<br>(-0.87)            | 0.0256 (1.13)              | (-0.0224)                     |
| Controls                  | Yes                        | Yes                           | Yes                        | Yes                           | Yes                        | Yes                           |
| Year fixed effects        | Yes                        | Yes                           | Yes                        | Yes                           | Yes                        | Yes                           |
| Industry fixed<br>effects | Yes                        | Yes                           | Yes                        | Yes                           | Yes                        | Yes                           |
| $R^2$                     | 0.0566                     | 0.0502                        | 0.0565                     | 0.0501                        | 0.0564                     | 0.0500                        |
| N                         | 5,563                      | 3,376                         | 5,563                      | 3,376                         | 5,563                      | 3,376                         |
| Panel B: all cash a       |                            |                               |                            |                               |                            |                               |
|                           | All cash<br>deal           | All stock<br>deal             | All cash<br>deal           | All stock<br>deal             | All cash<br>deal           | All stock<br>deal             |
|                           | (1)<br>CAR                 | (2)<br>CAR                    | (3)<br>CAR                 | (4)<br>CAR                    | (5)<br>CAR                 | (6)<br>CAR                    |
| PCO_OPT                   | -0.0060* (-1.88)           | -0.0020<br>(-0.30)            |                            |                               |                            |                               |
| NCO_OPT                   |                            | (                             | $-0.0005^{*}$<br>(-1.67)   | -0.0003<br>(-0.65)            |                            |                               |
| RCO_OPT                   |                            |                               |                            |                               | $-0.0056^{*}$<br>(-1.77)   | -0.0009<br>(-0.14)            |
| CONSTANT                  | 0.0021<br>(0.14)           | 0.2169***<br>(3.46)           | -0.0020<br>(-0.12)         | 0.2144***<br>(3.42)           | -0.0015<br>(-0.09)         | 0.2155***<br>(3.46)           |
| Controls                  | Yes                        | Yes                           | Yes                        | Yes                           | Yes                        | Yes                           |
| Year fixed effects        | Yes                        | Yes                           | Yes                        | Yes                           | Yes                        | Yes                           |
| Industry fixed<br>effects | Yes                        | Yes                           | Yes                        | Yes                           | Yes                        | Yes                           |
| R <sup>2</sup><br>N       | 0.0591<br>2,793            | 0.1106<br>1,258               | 0.0587<br>2,793            | 0.1108<br>1,258               | 0.0590<br>2,793            | 0.1106<br>1,258               |
| Panel C: related a        | nd unrelated deal          |                               |                            |                               |                            |                               |
|                           | Related deal (1)           | Unrelated<br>deal<br>(2)      | Related deal (3)           | Unrelated<br>deal<br>(4)      | Related deal<br>(5)        | Unrelated<br>deal<br>(6)      |
|                           | CÁR                        | CÁR                           | CÁR                        | CÁR                           | CÁR                        | CÁR                           |
| PCO_OPT                   | -0.0091***<br>(-3.28)      | -0.0057*<br>(-1.88)           | -                          | -                             | -                          | -                             |
| NCO_OPT                   | -                          | -                             | $-0.0009^{***}$<br>(-3.51) | $-0.0005^{**}$<br>(-1.96)     | -                          | -                             |
| RCO_OPT                   | -                          | -                             | -                          | -                             | -0.0082***<br>(-2.98)      | -0.0052*<br>(-1.73)           |
|                           |                            |                               |                            |                               |                            | (continue                     |

## Table 11. Continued

# China Accounting and Finance

| Panel C: related a                                    | and unrelated d            | eals                            |                            |                                 |                            |                                 | and Fina<br>Rev |
|---|----------------------------|---------------------------------|----------------------------|---------------------------------|----------------------------|---------------------------------|-----------------|
|   | Related deal<br>(1)<br>CAR | Unrelated<br>deal<br>(2)<br>CAR | Related deal<br>(3)<br>CAR | Unrelated<br>deal<br>(4)<br>CAR | Related deal<br>(5)<br>CAR | Unrelated<br>deal<br>(6)<br>CAR |                 |
| CONSTANT<br>Controls                                  | 0.0053<br>(0.25)           | 0.0084<br>(0.64)                | -0.0018<br>(-0.09)<br>Yes  | 0.0031<br>(0.23)<br>Yes         | 0.0002<br>(0.01)           | 0.0047<br>(0.36)<br>Yes         |                 |
| Year fixed<br>effects                                 | Yes<br>Yes                 | Yes<br>Yes                      | Yes                        | Yes                             | Yes<br>Yes                 | Yes                             |                 |
| Industry fixed<br>effects                             | Yes                        | Yes                             | Yes                        | Yes                             | Yes                        | Yes                             |                 |
| R <sup>2</sup><br>N                                   | 0.0566<br>4,715            | 0.0445<br>4,224                 | 0.0566<br>4,715            | 0.0444<br>4,224                 | 0.0562<br>4,715            | 0.0444<br>4,224                 |                 |
| Panel D: large an                                     | d small deals<br>Large d   | eal Small de                    | eal Large dea              | l Small deal                    | Large deal                 | Small deal                      |                 |
|   | (1)<br>CAR                 | (2)<br>CAR                      | (3)<br>CAR                 | (4)<br>CAR                      | (5)<br>CAR                 | (6)<br>CAR                      |                 |
| PCO_OPT   | -0.011<br>(-3.68)          |                                 |                            | _                               | -                          | -                               |                 |
| NCO_OPT   | -                          | -                               | -0.0011*<br>(-3.85)        | (-1.45)                         | -                          | _                               |                 |
| RCO_OPT   | _                          | -                               | _                          | _                               | -0.0106***<br>(-3.43)      | $^{*}$ -0.0041 (-1.52)          |                 |
| CONSTANT  | -0.027<br>(-1.01)          | ) (1.35)                        | -0.0369<br>(-1.31)         | 0.0208<br>(1.15)                | -0.0343<br>(-1.25)         | 0.0214<br>(1.19)                |                 |
| Controls<br>Year fixed effects<br>Industry fixed effe |                            | Yes<br>Yes<br>Yes               | Yes<br>Yes<br>Yes          | Yes<br>Yes<br>Yes               | Yes<br>Yes<br>Yes          | Yes<br>Yes<br>Yes               |                 |
| $R^2$<br>N  | 0.0847<br>4,431            | 0.0305<br>4,508                 | 0.0846<br>4,431            | 0.0303<br>4,508                 | 0.0843<br>4,431            | 0.0304<br>4,508                 |                 |
| Panel E: foreign o                                    | deals<br>Foreign           | Domestic                        | Foreign                    | Domestic                        | Foreign                    | Domestic                        |                 |
|   | deals<br>(1)<br>CAR        | deals<br>(2)<br>CAR             | deals<br>(3)<br>CAR        | deals<br>(4)<br>CAR             | deals<br>(5)<br>CAR        | deals<br>(6)<br>CAR             |                 |
| PCO_OPT   | $-0.0077^{**}$<br>(-2.08)  | $-0.0071^{***}$<br>(-3.02)      | _                          | -                               | -                          | -                               |                 |
| NCO_OPT   | -                          | -                               | $-0.0007^{**}$<br>(-2.17)  | $-0.0007^{***}$<br>(-3.11)      | -                          | -                               |                 |
| RCO_OPT   | -                          | -                               | _                          | _                               | -0.0071*<br>(-1.95)        | -0.0064***<br>(-2.73)           |                 |
| CONSTANT  | 0.0514**<br>(2.43)<br>Yes  | 0.0075<br>(0.40)<br>Yes         | 0.0454**<br>(2.17)<br>Yes  | 0.0015<br>(0.08)<br>Yes         | 0.0464**<br>(2.20)<br>Yes  | 0.0034<br>(0.18)<br>Yes         |                 |

(continued)

#### Table 11. Continued

| -                         | Foreign | Domestic | Foreign | Domestic | Foreign | Domestic |
|---------------------------|---------|----------|---------|----------|---------|----------|
|                           | deals   | deals    | deals   | deals    | deals   | deals    |
|                           | (1)     | (2)      | (3)     | (4)      | (5)     | (6)      |
|                           | CAR     | CAR      | CAR     | CAR      | CAR     | CAR      |
| Year fixed<br>effects     | Yes     | Yes      | Yes     | Yes      | Yes     | Yes      |
| Industry fixed<br>effects | Yes     | Yes      | Yes     | Yes      | Yes     | Yes      |
| $R^2$                     | 0.0537  | 0.0436   | 0.0535  | 0.0435   | 0.0535  | 0.0434   |
| N                         | 2,266   | 6,673    | 2,266   | 6,673    | 2,266   | 6,673    |

**Note(s):** This table presents the regression results of the effect of co-opted board on acquisition performance with other control variables for public and private deals in Panel A, all cash and all stock deals in Panel B, related and unrelated deals in Panel C and large and small deals in Panel D, and foreign and domestic deals in Panel E. Robust two-tailed *t*-statistics clustered by firm are presented in parentheses. The superscripts \*\*\*, \*\* and \* correspond to statistical significance at the 1%, 5% and 10% levels, respectively. We present the variable definitions in the Appendix

Source(s): Authors' own work

directors on acquisition performance is observed in the cash-only subsample, while it is insignificant in the stock-only subsample. In Panel C, we test whether there is an industry-specific influence on the relationship between board co-option and market reaction to the acquisition announcement. For this purpose, we split the sample into two groups: (1) acquirers and targets in the related industries and (2) acquirers and targets in the unrelated industries. We find that the coefficients of the board co-option variables remain positive and significant across both subsamples for the two measures of board co-option used. However, the coefficients in the related-industry subsample exhibit larger magnitudes and higher significance levels. This finding supports that the related-industry environments offer more opportunities for managers to engage in acquisitions that harm shareholders' wealth. Nonetheless, the reported evidence corroborates that our main results are robust to industry relatedness and that board co-option worsens acquisition performance irrespective of the relatedness in the industries in which acquirers and targets operate.

When we further investigate the influence of deal size by partitioning our sample into large and small deals in Panel D, we find that the higher the percentage and number of co-opted directors on the board, the more pronounced the negative effect it has on the market reaction on the announcement of a large deal. Finally, the sample is divided into two groups of foreign and domestic acquisitions in Panel E. The findings show that the association between board cooption and abnormal return remains negative and in a similar magnitude and significant range for both sub-samples. These findings reveal that high board co-option can be more detrimental through certain types of acquisitions. However, the fact that this analysis failed to uncover positive and significant coefficients for the board co-option measure in any of the subsamples analysed strongly reveals that the presence of co-opted directors on the board does not create value through any of these acquisition choices.

4.3.5 Acquisition efficiency. We now consider if board co-option negatively impacts shareholders' wealth through acquisition; that effect also should be revealed in acquisition efficiency. Therefore, we follow prior literature to investigate the impact of board co-option on a range of acquisition efficiency measures. First, we explore its effect on excessive premiums paid to targets, which is referred to as the main reason for acquisitions that do not enhance shareholders' wealth (Fishman, 1988; Flanagan & O'Shaughnessy, 2003; Mulherin & Boone, 2000; Roll, 1986). Second, we calculate the logarithm of days needed to complete the deal to include in our additional test because this variable represents the level of monitoring intensity

exhibited by the board while assessing the deal (Faleye *et al.*, 2011; Shams, Minnick, Khedmati, & Gunasekarage, 2024). Third, we consider that a partial acquisition can be a proxy for acquisition efficiency because it brings a number of benefits for an organisation, such as risk diversification, economies of scale advantage, promotion of organisational learning and reduced market competition (Nain & Wang, 2018). Finally, following Akhigbe, Martin, and Whyte (2007), we examine deal completion status because an acquisition is a significant corporate decision that consumes both human and financial resources, directly reflecting the effectiveness of the decision.

In this section, we investigate if board co-option is associated with any changes in the acquisition efficiency of acquirers by replacing the dependent variable in our baseline model with four proxies of acquisition efficiency, including bid premium (*BIDPREM*), time taken to complete an acquisition (*LOGDAYS*), the likelihood of conducting a partial acquisition, (*PARACQ*) and the likelihood of completing the deal (*COMPLETED*). We report the findings in Table 12. We find a significant positive association between board co-option and the bid premium paid in acquisitions. This finding corroborates our findings reported in the baseline model that board co-option reduces shareholders' wealth, not only through the market reaction surrounding the announcement date but also through the high premium paid to the target.

We also observe that board co-option variables generate negative and significant coefficients on all of the other three acquisition efficiency variables. In Panel B, firms of highly presented co-opted directors seem to place less emphasis on partial acquisitions, which can limit their strategic flexibility. This phenomenon may arise because co-opted directors often exhibit loyalty to the CEO, potentially leading to a preference for larger, more transformative deals rather than incremental acquisitions that diversify risk (Coles *et al.*, 2014; Hermalin & Weisbach, 1998) while discouraging the pursuit of partial acquisitions that might provide valuable benefits, including risk diversification and opportunities for gradual growth (Nain & Wang, 2018).

Panel C reports that managers in highly co-opted boards spend less time to complete the deal. While a shorter deal completion time might superficially suggest operational efficiency, we argue that this finding actually points to efficiency reduction in acquisitions, as faster deal completion may indicate less rigorous scrutiny by the board. Previous literature emphasises that effective monitoring often requires sufficient time to assess the strategic fit, conduct due diligence and ensure shareholder value (Faleye *et al.*, 2011). When co-opted directors, who are more aligned with the CEO, reduce this scrutiny, it can result in a quicker but less thorough process, potentially leading to suboptimal acquisition outcomes in the long term (Hermalin & Weisbach, 1998; Coles *et al.*, 2014). Therefore, we interpret this faster deal completion as a sign of weakened governance and a reduction in the quality of decision-making, rather than a true improvement in efficiency.

Additionally, Panel D reports that firms with co-opted directors appear less likely to complete acquisition deals, potentially due to weakened governance and oversight. The loyalty of co-opted directors to the CEO may lead to a lack of critical evaluation of proposed acquisitions. As a result, the propensity to finalise acquisitions diminishes, reflecting a governance structure that prioritises CEO preferences over strategic opportunities, ultimately impacting firm performance and shareholder value. Collectively, these findings indicate that board co-option is associated with a reduction in acquisition efficiency.

4.3.6 Change in performance. Thus far, we have reported that board co-option reduces acquisition performance. We next explore what the effect on performance is if a highly co-opted board executes acquisitions inefficiently and the market reacts unfavourably to the deal announcement. To execute this analysis, we calculate the change in performance of the bidders, measured by the return of assets and Tobin's Q and re-estimate equation (1) replacing announcement returns with change in performance variables. The sample of these analyses is reduced to 7,932 and 7,776 observations, respectively. As presented in Table 13, we find that the coefficients of the *PCO\_OPT*, *NCO\_OPT* and *RCO\_OPT* variables are negative and significant in both models estimated, implying that co-opted boards make acquisitions that go

China Accounting and Finance Review

Panel A: bid premium Panel B: partial deals (1)(3)(4) (6) (2)(5)BÍDPREM BÍDPREM BIDPREM PARACQ PARACQ PARACQ 0.6112\*\*\* PCO\_OPT -0.1926\*\*(3.42)(-2.09)0.0586\*\*\* NCO\_OPT -0.0158\*(3.18)(-1.77)RCO\_OPT 0.6051\*\*\* -0.1994 \*\*(3.38)(-2.18)CONSTANT -1.7518-1.2361-2.2066\*\*\*-2.3476\*\*\* -2.3359\*\*\* -1.3581(-2.76)(-1.14)(-0.79)(-0.88)(-2.62)(-2.82)Controls Yes Yes Yes Yes Yes Yes Year fixed effects Yes Yes Yes Yes Yes Yes Industry fixed Yes Yes Yes Yes Yes Yes effects *R*<sup>2</sup>/Wald-chi2 0.4041 0.4040 0.4041 1159.9506 1158.9619 1160.2459 8,939 Ν 8,939 8,939 8,939 8,939 8,939 Panel C: days to completion Panel D: completed deals (3)(6)(1)(2)(4) (5) LOGDAYS LOGDAYS LOGDAYS COMPLETED COMPLETED COMPLETED PCO\_OPT -0.2044 \*\*\*-0.3283\*\*\*(-2.69)(-2.96)NCO\_OPT -0.0226\*\*\* -0.0238\*\*(-2.25)(-3.09)RCO\_OPT -0.2024\*\*\*-0.3186\*\*\*(-2.68)(-2.91)1.6797\* CONSTANT -0.75471.4604 -0.5520-0.68371.4691 (-0.74)(-1.00)(-0.91)(1.83)(1.53)(1.59)Controls Yes Yes Yes Yes Yes Yes Year fixed Yes Yes Yes Yes Yes Yes effects Industry Yes Yes Yes Yes Yes Yes fixed effects R<sup>2</sup>/Wald-0.2778 0.2781 0.2778 1544.1527 1542.2412 1545.0912 chi2 Ν 7.099 7.099 7.099 8.939 8.939 8.939

**Note(s):** This table presents the regression results of co-opted board on acquisition efficiency with other control variables. Robust two-tailed t-statistics clustered by firm are presented in parentheses. The superscripts \*\*\*, \*\* and \* correspond to statistical significance at the 1%, 5% and 10% levels, respectively. We present the variable definitions in the Appendix

Source(s): Authors' own work

against shareholders' interests, leading to worse financial performance of the acquirer in the consecutive year.

## 5. Conclusion

This study examines the impact of board co-option on bidder announcement returns using a sample of US M&As from 1999 to 2019. The results show that bidder cumulative abnormal returns are lower when the board is co-opted. It supports the board's monitoring role and suggests that the monitoring environment is weaker under co-opted boards because of their loyalty to CEOs. We also show that the negative association between board co-option and

#### **Table 12.** Co-opted board and market reactions to acquisition announcements: acquisition efficiency

| 10010 101 00 0            | pice board and            | r enange in pe           | inominance and       | marnet variation    | •                        |                      | and Einance           |
|---------------------------|---------------------------|--------------------------|----------------------|---------------------|--------------------------|----------------------|-----------------------|
|                           | (1)<br>CH1ROA             | (2)<br>CH1ROA            | (3)<br>CH1ROA        | (4)<br>CH1TOBINQ    | (5)<br>CH1TOBINQ         | (6)<br>CH1TOBINQ     | and Finance<br>Review |
| PCO_OPT                   | $-0.0596^{**}$<br>(-2.27) | _                        | -                    | -0.2364*<br>(-1.95) | -                        | -                    |                       |
| NCO_OPT                   | _                         | $-0.0032^{*}$<br>(-1.71) | -                    | _                   | $-0.0163^{*}$<br>(-1.74) | -                    |                       |
| RCO_OPT                   | -                         | _                        | -0.0585**<br>(-2.20) | -                   |                          | -0.2437**<br>(-2.00) |                       |
| CONSTANT                  | 0.1935<br>(1.31)          | 0.1667<br>(1.07)         | 0.1555<br>(1.00)     | -0.6816<br>(-0.39)  | -0.8119<br>(-0.47)       | -0.8395<br>(-0.49)   |                       |
| Controls                  | Yes                       | Yes                      | Yes                  | Yes                 | Yes                      | Yes                  |                       |
| Year fixed<br>effects     | Yes                       | Yes                      | Yes                  | Yes                 | Yes                      | Yes                  |                       |
| Industry fixed<br>effects | Yes                       | Yes                      | Yes                  | Yes                 | Yes                      | Yes                  |                       |
| $R^2$<br>N                | 0.0184<br>7,932           | 0.0178<br>7,932          | 0.0184<br>7,932      | 0.1546<br>7,776     | 0.1543<br>7,776          | 0.1546<br>7,776      |                       |

China Accounting

Table 13. Co-opted board and change in performance and market valuation

**Note(s):** This table presents the regression results of co-opted board on firms' change in performance (ROA) and market valuation (Tobin Q) with other control variables. Robust two-tailed *t*-statistics clustered by firm are presented in parentheses. The superscripts \*\*\*, \*\* and \* correspond to statistical significance at the 1%, 5% and 10% levels, respectively. See the Appendix for the variable definitions **Source(s):** Authors' own work

\_\_\_\_\_

bidder returns is less pronounced for firms led by CEOs who are younger, those who are less risk-taking, have longer tenures and are male, compared to their counterparts. We further find that board co-option reduces acquisition efficiency and leads to worse performance. Our results hold after controlling for board characteristics and addressing endogeneity-related issues.

We contribute to the M&A literature and suggest that board co-option is detrimental to bidder shareholders, thereby offering important implications for regulators and policymakers. Specifically, we highlight that poor board monitoring, exacerbated by the presence of co-opted directors, can adversely influence shareholder returns in acquisition announcements. This weakens the board's ability to act independently, allowing CEOs to pursue acquisitions that may not align with shareholder interests.

For regulators and policymakers, our results highlight the ongoing need for stringent governance practices that promote truly independent board oversight, especially in the post-Sarbanes-Oxley era. Although the Sarbanes-Oxley Act aimed to improve corporate governance, our study suggests that CEO influence over board members can persist even in regulated environments. Policymakers could consider reinforcing rules to limit CEO involvement in the selection of directors, ensuring that boards remain effective in protecting shareholder value.

Furthermore, corporate governance reforms could focus on mechanisms that prevent CEO overreach and enhance board independence. Requiring staggered board elections, independent nomination committees or stricter limits on tenure and social ties between the CEO and board members might improve governance quality. By implementing such measures, regulators can safeguard firms from the adverse effects of board co-option, ensuring that corporate decisions, particularly in M&A, are made with greater transparency and accountability.

Nonetheless, our study is subject to certain limitations that open avenues for further work on board co-option in M&As. Future research can examine how board co-option influences synergistic gains in the international takeover market. Our findings can stimulate future work to investigate how co-opted boards affect other non-financial deal consequences, such as CAFR employment and culture. We considered publicly listed firms to test the association between board co-option and bidder returns and similar work can be expanded to private firms.

#### Notes

- Although the Sarbanes Oxley Act of 2002, NASDAQ and NYSE endorsed listing essentials to diminish CEOs' undeviating influence on the recruiting process, CEOs continue to exert their influence on this process.
- 2. Following Amel-Zadeh and Meeks (2019) and Fich and Nguyen (2020), first we exclude share buyback, share repurchase, acquisition of assets and exchange offers and transactions with less than 50% shares acquired.
- 3. We use the Standard Industry Classification (SIC) for assigning companies into 48 industries.
- 4. The "Others" industry category reported in Table 1 comprised of remaining 38 industry classification.

#### References

- Akdogu, E. (2011). Value-maximizing managers, value-increasing mergers. *Journal of Financial and Quantitative Analysis*, 46(1), 83–110. doi: 10.1017/s002210901000075x.
- Akhigbe, A., Martin, A. D., & Whyte, A. M. (2007). Partial acquisitions, the acquisition probability hypothesis, and the abnormal returns to partial targets. *Journal of Banking and Finance*, 31(10), 3080–3101. doi: 10.1016/j.jbankfin.2006.11.015.
- Alexandridis, G., Fuller, K. P., Terhaar, L., & Travlos, N. G. (2013). Deal size, acquisition premia and shareholder gains. *Journal of Corporate Finance*, 20, 1–13. doi: 10.1016/j.jcorpfin.2012.10.006.
- Amel-Zadeh, A., & Meeks, G. (2019). Bidder earnings forecasts in mergers and acquisitions. *Journal of Corporate Finance*, 58, 373–392. doi: 10.1016/j.jcorpfin.2019.06.002.
- Andrade, G., Mitchell, M., & Stafford, E. (2001). New evidence and perspectives on mergers. *The Journal of Economic Perspectives*, 15(2), 103–120. doi: 10.1257/jep.15.2.103.
- Armstrong, C. S., Balakrishnan, K., & Cohen, D. (2012). Corporate governance and the information environment: Evidence from state antitakeover laws. *Journal of Accounting and Economics*, 53 (1-2), 185–204. doi: 10.1016/j.jacceco.2011.06.005.
- Baghdadi, G. A., Nguyen, L. H., & Podolski, E. J. (2020). Board co-option and default risk. *Journal of Corporate Finance*, 64, 101703. doi: 10.1016/j.jcorpfin.2020.101703.
- Bange, M. M., & Mazzeo, M. A. (2004). Board composition, board effectiveness, and the observed form of takeover bids. *Review of Financial Studies*, 17(4), 1185–1215. doi: 10.1093/rfs/hhh001.
- Beasley, M. S., Carcello, J. V., Hermanson, D. R., & Lapides, P. D. (2000). Fraudulent financial reporting: Consideration of industry traits and corporate governance mechanisms. *Accounting Horizons*, 14(4), 441–454. doi: 10.2308/acch.2000.14.4.441.
- Bebchuk, L. A., Cohen, A., & Wang, C. C. (2013). Learning and the disappearing association between governance and returns. *Journal of Financial Economics*, 108(2), 323–348. doi: 10.1016/ j.jfineco.2012.10.004.
- Belloni, A., Chernozhukov, V., & Hansen, C. (2014). High-dimensional methods and inference on structural and treatment effects. *The Journal of Economic Perspectives*, 28(2), 29–50. doi: 10.1257/jep.28.2.29.
- Belloni, A., Chernozhukov, V., Fernández-Val, I., & Hansen, C. (2017). Program evaluation and causal inference with high-dimensional data. *Econometrica*, 85(1), 233–298. doi: 10.3982/ecta12723.
- Bertrand, O., & Betschinger, M.-A. (2012). Performance of domestic and cross-border acquisitions: Empirical evidence from Russian acquirers. *Journal of Comparative Economics*, *40*(3), 413–437. doi: 10.1016/j.jce.2011.11.003.
- Bertrand, M., & Schoar, A. (2003). Managing with style: The effect of managers on firm policies. *Quarterly Journal of Economics*, 118(4), 1169–1208. doi: 10.1162/003355303322552775.

- Black, B. S., Jang, H., & Kim, W. (2006). Does corporate governance predict firms' market values? Evidence from Korea. *Journal of Law, Economics, and Organization*, 22(2), 366–413. doi: 10.1093/jleo/ewj018.
- Bradley, M., Desai, A., & Kim, E. H. (1988). Synergistic gains from corporate acquisitions and their division between the stockholders of target and acquiring firms. *Journal of Financial Economics*, 21(1), 3–40. doi: 10.1016/0304-405x(88)90030-x.
- Brennan, M. J., & Subrahmanyam, A. (1995). Investment analysis and price formation in securities markets. *Journal of Financial Economics*, 38(3), 361–381. doi: 10.1016/0304-405x(94)00811-e.
- Brown, S. J., & Warner, J. B. (1985). Using daily stock returns: The case of event studies. *Journal of Financial Economics*, 14(1), 3–31. doi: 10.1016/0304-405x(85)90042-x.
- Byrd, J. W., & Hickman, K. A. (1992). Do outside directors monitor managers? Evidence from tender offer bids. *Journal of Financial Economics*, 32(2), 195–221. doi: 10.1016/0304-405x(92) 90018-s.
- Chang, S. (1998). Takeovers of privately held targets, methods of payment, and bidder returns. *The Journal of Finance*, 53(2), 773–784. doi: 10.1111/0022-1082.315138.
- Chiu, P. C., Teoh, S. H., & Tian, F. (2013). Board interlocks and earnings management contagion. *The Accounting Review*, 88(3), 915–944. doi: 10.2308/accr-50369.
- Coles, J. L., Daniel, N. D., & Naveen, L. (2014). Co-opted boards. *Review of Financial Studies*, 27(6), 1751–1796. doi: 10.1093/rfs/hhu011.
- Cook, D. O., Chowdhury, J., & Zhang, W. (2023). Director optimism and CEO equity compensation. *Journal of Empirical Finance*, 72, 143–162. doi: 10.1016/j.jempfin.2023.03.010.
- Cotter, J. F., Shivdasani, A., & Zenner, M. (1997). Do independent directors enhance target shareholder wealth during tender offers. *Journal of Financial Economics*, *43*(2), 195–218. doi: 10.1016/s0304-405x(96)00886-0.
- Cumming, D., Leung, T. Y., & Rui, O. (2015). Gender diversity and securities fraud. Academy of Management Journal, 58(5), 1572–1593. doi: 10.5465/amj.2013.0750.
- Dahlerup, D. (2006). The story of the theory of critical mass. *Politics and Gender*, *2*(4), 511–522. doi: 10.1017/s1743923x0624114x.
- Danbolt, J., & Maciver, G. (2012). Cross-border versus domestic acquisitions and the impact on shareholder wealth. *Journal of Business Finance and Accounting*, *39*(7-8), 1028–1067. doi: 10.1111/j.1468-5957.2012.02294.x.
- Datta, S., Iskandar-Datta, M., & Raman, K. (2001). Executive compensation and corporate acquisition decisions. *The Journal of Finance*, 56(6), 2299–2336. doi: 10.1111/0022-1082.00406.
- Davidson, R., Goodwin-Stewart, J., & Kent, P. (2005). Internal governance structures and earnings management. Accounting and Finance, 45(2), 241–267. doi: 10.1111/j.1467-629x.2004.00132.x.
- Davidson, R., Dey, A., & Smith, A. (2015). Executives' 'off-the-job' behavior, corporate culture, and financial reporting risk. *Journal of Financial Economics*, 117(1), 5–28. doi: 10.1016/ j.jfineco.2013.07.004.
- De, S., Fedenia, M., & Triantis, A. J. (1996). Effects of competition on bidder returns. Journal of Corporate Finance, 2(3), 261–282. doi: 10.1016/0929-1199(95)00011-9.
- De Franco, G., Hou, Y., Ma, M. S. (2022). Do firms mimic industry leaders' accounting? Evidence from financial statement comparability. SSRN 3310219.
- Doidge, C., Karolyi, G. A., & Stulz, R. M. (2007). Why do countries matter so much for corporate governance ?. Journal of Financial Economics, 86, 1–39. doi: 10.1016/j.jfineco.2006.09.002.
- Edmans, A., & Holderness, C. G. (2017). Blockholders: A survey of theory and evidence. In *The Handbook of the Economics of Corporate Governance*, *1*, 541–636.
- Faleye, O., Hoitash, R., & Hoitash, U. (2011). The costs of intense board monitoring. Journal of Financial Economics, 101(1), 160–181. doi: 10.1016/j.jfineco.2011.02.010.
- Fama, E. F., Fisher, L., Jensen, M. C., & Roll, R. (1969). The adjustment of stock prices to new information. *International Economic Review*, 10(1), 1–21. doi: 10.2307/2525569.

China Accounting and Finance Review

- Fich, E. M., & Nguyen, T. (2020). The value of CEOs' supply chain experience: Evidence from Mergers and Acquisitions. *Journal of Corporate Finance*, 60, 101525. doi: 10.1016/ j.jcorpfin.2019.101525.
  - Finkelstein, S. (1992). Power in top management teams: Dimensions, measurement, and validation. *Academy of Management Journal*, *35*(3), 505–538. doi: 10.2307/256485.
  - Finkelstein, S., & Hambrick, D. C. (1989). Chief executive compensation: A study of the intersection of markets and political processes. *Strategic Management Journal*, 10(2), 121–134. doi: 10.1002/smj.4250100203.
  - Finkelstein, S., Hambrick, D. C., & Cannella, A. A. (2009). *Strategic leadership: Theory and research* on executives, top management teams, and boards. New York, NY: Oxford University Press.
  - Fishman, M. J. (1988). A theory of preemptive takeover bidding. *The RAND Journal of Economics*, 19 (1), 88–101. doi: 10.2307/2555399.
  - Flanagan, D. J., & O'Shaughnessy, K. C. (2003). Core-related acquisitions, multiple bidders and tender offer premiums. *Journal of Business Research*, 56(8), 573–585. doi: 10.1016/s0148-2963(01) 00269-7.
  - Fracassi, C., & Tate, G. (2012). External networking and internal firm governance. *The Journal of Finance*, 67(1), 153–194. doi: 10.1111/j.1540-6261.2011.01706.x.
  - Frye, M. B., & Pham, D. T. (2018). CEO gender and corporate board structures. *The Quarterly Review* of *Economics and Finance*, 69, 110–124. doi: 10.1016/j.qref.2017.12.002.
  - Graham, J. R., Kim, H., & Leary, M. (2017). CEO power and board dynamics. In *Paper presented at the IDC summer finance conference*, Tuck.
  - Grossman, S. J., & Hart, O. D. (1980). Takeover bids, the free-rider problem, and the theory of the corporation. *The Bell Journal of Economics*, *11*(1), 42–64. doi: 10.2307/3003400.
  - Gujarati, D. N., & Porter, D. C. (2009). Basic econometrics (5th ed.). New York: McGraw Hill.
  - Gunasekarage, A., Luong, H., & Truong, T. T. (2020). Growth and market share matrix, CEO power, and firm performance. *Pacific-Basin Finance Journal*, 59, 101257. doi: 10.1016/ j.pacfin.2019.101257.
  - Hagendorff, J., & Vallascas, F. (2011). CEO pay incentives and risk-taking: Evidence from bank acquisitions. *Journal of Corporate Finance*, 17(4), 1078–1095. doi: 10.1016/ j.jcorpfin.2011.04.009.
  - Hermalin, B. E., & Weisbach, M. S. (1998). Endogenously chosen boards of directors and their monitoring of the CEO. American Economic Review, 88, 96–118.
  - Hill, C. W. L., & Jones, T. M. (1992). Stakeholder-agency theory. *Journal of Management Studies*, 29 (2), 131–154. doi: 10.1111/j.1467-6486.1992.tb00657.x.
  - Hoang, H., & Antoncic, B. (2003). Network-based research in entrepreneurship: A critical review. *Journal of Business Venturing*, 18(2), 165–187. doi: 10.1016/s0883-9026(02)00081-2.
  - Huang, H., Han, S. H., & Cho, K. (2021). Co-opted boards, social capital, and risk-taking. *Finance Research Letters*, 38, 101535. doi: 10.1016/j.frl.2020.101535.
  - Humphery-Jenner, M. L., & Powell, R. G. (2011). Firm size, takeover profitability, and the effectiveness of the market for corporate control: Does the absence of anti-takeover provisions make a difference?. *Journal of Corporate Finance*, *17*(3), 418–437. doi: 10.1016/ j.jcorpfin.2011.01.002.
  - Humphery-Jenner, M., Islam, E., Rahman, L., & Suchard, J. A. (2022). Powerful CEOs and corporate governance. *Journal of Empirical Legal Studies*, 19(1), 135–188. doi: 10.1111/jels.12305.
  - Hussain, T., & Shams, S. (2022). Pre-deal differences in corporate social responsibility and acquisition performance. *International Review of Financial Analysis*, 81, 1–17. doi: 10.1016/ j.irfa.2022.102083.
  - Hwang, B. -H., & Kim, S. (2009). It pays to have friends. *Journal of Financial Economics*, 93(1), 138–158. doi: 10.1016/j.jfineco.2008.07.005.

| Jain, T., & Zaman, R. (2020). When boards matter: The case of corporate social irresponsibility. <i>British Journal of Management</i> , <i>31</i> (2), 365–386. doi: 10.1111/1467-8551.12376.   | China Accounting<br>and Finance |
|---|---------------------------------|
| Jensen, M. (1986). Agency costs of free cash flow, corporate finance, and takeovers. <i>The American Economic Review</i> , <i>76</i> (2), 323.  | Review                          |
| Jensen, M., & Zajac, E. (2004). Corporate elites and corporate strategy: How demographic preferences and structural position shape the scope of the firm. <i>Strategic Management Journal</i> , <i>25</i> (6), 507–524. doi: 10.1002/smj.393.         |                                 |
| Jiraporn, P., & Lee, S. M. (2018). Do co-opted directors influence dividend policy?. <i>Financial Management</i> , <i>47</i> (2), 349–381. doi: 10.1111/fima.12196.   |                                 |
| Kao, E. H., Huang, HC., Fung, HG., & Liu, X. (2020). Co-Opted directors, gender diversity, and<br>crash risk: Evidence from China. <i>Review of Quantitative Finance and Accounting</i> , 55(2),<br>461–500. doi: 10.1007/s11156-019-00850-3.         |                                 |
| Kirsch, A. (2018). The gender composition of corporate boards: A review and research agenda. <i>The Leadership Quarterly</i> , 29(2), 346–364. doi: 10.1016/j.leaqua.2017.06.001.   |                                 |
| Larcker, D. F., & Rusticus, T. O. (2010). On the use of instrumental variables in accounting research. <i>Journal of Accounting and Economics</i> , 49(3), 186–205. doi: 10.1016/j.jacceco.2009.11.004.   |                                 |
| Lartey, T., Danso, A., & Boateng, A. (2021). Co-opted boards and capital structure dynamics.<br>International Review of Financial Analysis, 77, 101824. doi: 10.1016/j.irfa.2021.101824.  |                                 |
| Lee, S. M., Jiraporn, P., Kim, Y. S., & Park, K. J. (2021). Do co-opted directors influence corporate risk-taking and credit ratings?. <i>The Quarterly Review of Economics and Finance</i> , <i>79</i> , 330–344. doi: 10.1016/j.qref.2020.07.003.   |                                 |
| Levi, M., Li, K., & Zhang, F. (2014). Director gender and mergers and acquisitions. <i>Journal of Corporate Finance</i> , 28, 185–200. doi: 10.1016/j.jcorpfin.2013.11.005.   |                                 |
| Li, Y., Gong, M., Zhang, X. Y., & Koh, L. (2018). The impact of environmental, social, and governance disclosure on firm value: The role of CEO power. <i>The British Accounting Review</i> , <i>50</i> (1), 60–75. doi: 10.1016/j.bar.2017.09.007.   |                                 |
| Lim, J., Do, V., & Vu, T. (2020). Co-opted directors, covenant intensity, and covenant violations.<br><i>Journal of Corporate Finance</i> , 64, 101628. doi: 10.1016/j.jcorpfin.2020.101628.  |                                 |
| Luong, H., Gunasekarage, A., & Shams, S. (2021). CEO pay slice and acquisitions in Australia: The<br>role of tournament incentives. <i>International Journal of Managerial Finance</i> , 18(5), 833–868.<br>doi: 10.1108/IJMF-06-2021-0292.           |                                 |
| Luong, H., Khedmati, M., Nguyen, L. A., Nigmonov, A., Ovi, N. Z., & Shams, S. (2023). CEO-director ties and board gender diversity: US evidence. <i>Journal of Behavioral and Experimental Finance</i> , 40, 100861. doi: 10.1016/j.jbef.2023.100861. |                                 |
| Luong, H., Minnick, K., Rivolta, M. L., & Sham, S. (2024). <i>CEO connectedness and firm transparency</i> .<br>European Financial Management.   |                                 |
| Masulis, R. W., Wang, C., & Xie, F. E. I. (2007). Corporate governance and acquirer returns. <i>The Journal of Finance</i> , 62(4), 1851–1889. doi: 10.1111/j.1540-6261.2007.01259.x.   |                                 |
| Moeller, S. B., Schlingemann, F. P., & Stulz, R. M. (2004). Firm size and the gains from acquisitions. <i>Journal of Financial Economics</i> , <i>73</i> (2), 201–228. doi: 10.1016/j.jfineco.2003.07.002.  |                                 |
| Morck, R., Shleifer, A., & Vishny, R. W. (1990). Do managerial objectives drive bad acquisitions?. <i>The Journal of Finance</i> , 45(1), 31–48. doi: 10.1111/j.1540-6261.1990.tb05079.x.   |                                 |
| Morse, A., Nanda, V., & Seru, A. (2011). Are incentive contracts rigged by powerful CEOs?. <i>The Journal of Finance</i> , 66(5), 1779–1821. doi: 10.1111/j.1540-6261.2011.01687.x.   |                                 |
| Mulherin, J. H., & Boone, A. L. (2000). Comparing acquisitions and divestitures. <i>Journal of Corporate Finance</i> , 6(2), 117–139. doi: 10.1016/s0929-1199(00)00010-9.   |                                 |
| Nain, A., & Wang, Y. (2018). The product market impact of minority stake acquisitions. <i>Management Science</i> , <i>64</i> (2), 825–844. doi: 10.1287/mnsc.2016.2575.   |                                 |
| Officer, M. S. (2007). The price of corporate liquidity: Acquisition discounts for unlisted targets. <i>Journal of Financial Economics</i> , 83(3), 571–598. doi: 10.1016/j.jfineco.2006.01.004.  |                                 |

- Oliver, P. E., & Marwell, G. (2001). Whatever happened to critical mass theory? A retrospective and assessment. *Sociological Theory*, 19(3), 292–311. doi: 10.1111/0735-2751.00142.
- Pablo, E. (2013). Cross-border diversification through M&s in Latin America. *Journal of Business Research*, 66(3), 425–430. doi: 10.1016/j.jbusres.2012.04.009.
- Rahman, D., Malik, I., Ali, S., & Iqbal, J. (2021). Do co-opted boards increase insider profitability?. *Journal of Contemporary Accounting and Economics*, 17(3), 100265. doi: 10.1016/j.jcae.2021.100265.
- Renneboog, L., & Vansteenkiste, C. (2019). Failure and success in mergers and acquisitions. *Journal of Corporate Finance*, 58, 650–699. doi: 10.1016/j.jcorpfin.2019.07.010.
- Roll, R. (1986). The hubris hypothesis of corporate takeovers. *Journal of Business*, 59(2), 197. doi: 10.1086/296325.
- Rosen, R. (2006). Merger momentum and investor sentiment: The stock market reaction to merger announcements. *Journal of Business*, 79(2), 987–1017. doi: 10.1086/499146.
- Rosenbaum, P. R., & Rubin, D. B. (1983). The central role of the propensity score in observational studies for causal effects. *Biometrika*, *70*(1), 41–55. doi: 10.2307/2335942.
- Schoenberg, R. (2000). The influence of cultural compatibility within cross-border acquisitions: A review. In *Advances in mergers and acquisitions*.
- Schwert, G. W. (2000). Hostility in takeovers: In the eyes of the beholder?. *The Journal of Finance*, 55 (6), 2599–2640. doi: 10.1111/0022-1082.00301.
- Shams, S., Minnick, K., Khedmati, M., & Gunasekarage, A. (2024). Gender diversity and acquisitions: How female directors add value in acquisition decisions. *Journal of Business Finance and Accounting*, 51(1-2), 113–147. doi: 10.1111/jbfa.12688.
- Shleifer, A., & Vishny, R. (2003). Stock market driven acquisitions. *Journal of Financial Economics*, 70(3), 295–311. doi: 10.1016/s0304-405x(03)00211-3.
- Starks, L. T., & Wei, K. D. (2013). Cross-Border mergers and differences in corporate governance. *International Review of Finance*, 13(3), 265–297. doi: 10.1111/irfi.12008.
- Tibshirani, R. (1996). Regression shrinkage and selection via the lasso. *Journal of the Royal Statistical Society: Series B*, 58(1), 267–288. doi: 10.1111/j.2517-6161.1996.tb02080.x.
- Travlos, N. G. (1987). Corporate takeover bids, methods of payment, and bidding firms' stock returns. *The Journal of Finance*, *42*(4), 943–963. doi: 10.2307/2328300.
- Ulupinar, B. (2018). The effect of managerial entrenchment on analyst bias. *Global Finance Journal*, 37, 25–38. doi: 10.1016/j.gfj.2018.04.001.
- Wintoki, M. B., & Xi, Y. (2019). Friendly directors and the cost of regulatory compliance. *Journal of Corporate Finance*, 58, 112–141. doi: 10.1016/j.jcorpfin.2019.04.011.
- Wintoki, M. B., Linck, J. S., & Netter, J. M. (2012). Endogeneity and the dynamics of internal corporate governance. *Journal of Financial Economics*, 105(3), 581–606. doi: 10.1016/j.jfineco.2012.03.005.
- Wooldridge, J. M. (2010). Econometric analysis of cross section and panel data. Cambridge: MIT Press.
- Wright, P., Kroll, M., & Elenkov, D. (2002). Acquisition returns, increase in firm size, and chief executive officer compensation: The moderating role of monitoring. *Academy of Management Journal*, 45(3), 599–608. doi: 10.2307/3069384.
- Yim, S. (2013). The acquisitiveness of youth: CEO age and acquisition behavior. *Journal of Financial Economics*, *108*(1), 250–273. doi: 10.1016/j.jfineco.2012.11.003.
- Zaman, R., Atawnah, N., Baghdadi, G. A., & Liu, J. (2021). Fiduciary duty or loyalty? Evidence from co-opted boards and corporate misconduct. *Journal of Corporate Finance*, 70, 102066. doi: 10.1016/j.jcorpfin.2021.102066.
- Zattoni, A., Dedoulis, E., Leventis, S., & Ees, H. V. (2020). Corporate governance and institutions a review and research agenda. *Corporate Governance: An International Review*, *28*(6), 465–487. doi: 10.1111/corg.12347.
- Zhou, B., Dutta, S., & Zhu, P. (2020). CEO tenure and mergers and acquisitions. *Finance Research Letters*, 34, 101277. doi: 10.1016/j.frl.2019.08.025.

# Appendix

China Accounting and Finance Review

Table A1. Definitions of variables used in main models

| Variable                                    | Definition   |
|---|--|
| Dependent and indep                         | endent variables   |
| PCO_OPT                                     | Number of co-opted directors divided by number of directors on the board. A director is considered to be a co-opted director if he/she is appointed after the incumbent CEO assumes office   |
| NCO_OPT                                     | The distance between the number of co-opted directors on the board and the number of directors on the board  |
| RCO_OPT<br>CAR                              | The residual from a regression of PCO_OPT on CEO tenure<br>Following Chang (1998), Moeller <i>et al.</i> (2004). and Masulis <i>et al.</i> (2007), we calculate<br>the three-day cumulative abnormal return earned by the acquirer during the M&A<br>announcement date, using the market model over a period of 200 days (-210, -11)<br>preceding the announcement date. We exclude the 10-day window immediately prior to<br>the acquisition announcement period from the estimation period because it is common in<br>acquisition events that the information is leaked to the capital market well before the<br>actual announcement |
| Firm characteristics                        |  |
| SIZE<br>LEV<br>CASH<br>TOBINQ<br>ROA        | The natural logarithm of acquiring firm's total assets<br>Acquiring firm's total debt divided by total assets<br>Firm's total cash and equivalent divided by total assets<br>Acquiring firm's market capitalisation plus total liabilities divided by total assets<br>Acquiring firm's earnings before interest, depreciation and amortisation, divided by total<br>book assets of the acquiring firms   |
| Corporate governance                        |  |
| CEO_DUALITY                                 | Indicator variable that takes the value of one if both CEO and chair positions are held by the same person, and zero otherwise   |
| LNCEO_TENURE<br>LNBOARD_SIZE<br>BOARD_INDEP | The natural logarithm of CEO tenure<br>The natural logarithm of total number of directors of the firm<br>The percentage of independent directors on the board  |
| LNANALYSTS                                  | The natural logarithm of firm's total number of analysts following   |
| Deal characteristics<br>RELATIVESIZE        | Transaction value reported by SDC divided by the market value of the acquirer one month prior to the acquisition announcement  |
| ALLCASH                                     | A dummy variable that equals to one if the deal is fully financed by cash, and zero otherwise  |
| ALLSTOCK                                    | A dummy variable that equals to one if the deal is fully financed by stock, and zero otherwise   |
| RELATEDID                                   | A dummy variable that equals to one if bidder and target are from the same industries, zero otherwise  |
| TENDEROFFER<br>COMPETINGBID                 | A dummy variable that takes the value of one if the deal is tender, zero otherwise<br>A dummy variable that takes the value of one if there are at least two acquirers making an<br>offer to a target, zero otherwise  |
| Other variables<br>Entrenchment index       |  |
| Institutional ownersh<br>YOUNGCEO           | ip<br>A dummy variable that takes the value of one if the CEO is younger than 65 years old,<br>zero otherwise  |
| CEO VEGA<br>CEO TENURE                      | The natural logarithm of the CEO's vega<br>The natural logarithm of the CEO's tenure   |
|   | (continued)  |

| Variable              | Definition   |
|-----------------------|--|
| CEO GENDER<br>BIDPREM | A dummy variable equal to 1 if the CEO is female, and zero otherwise<br>For public targets: The difference between the deal value and the market capitalisation<br>one month before the deal announcement<br>For private targets: The average bid premium paid to public targets in a given industry<br>and year |
| PARACQ                | A dummy variable equals to one if the bidder acquires less than 100%, and zero otherwise   |
| LOGDAYS               | The natural logarithm of the number of days between bid announcement and deal execution  |
| COMPLETED             | A dummy variable equals to one if the deal is completed, and zero otherwise  |

## **Corresponding author**

Hoa Luong can be contacted at: hoa.luong@otago.ac.nz